# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM S-8 REGISTRATION STATEMENT

*under THE SECURITIES ACT OF 1933* 

# PARAMOUNT GLOBAL

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

1515 Broadway, New York, New York (Address of Principal Executive Offices) 04-2949533 (I.R.S. Employer Identification No.)

> 10036 (Zip Code)

Paramount Global Excess 401(k) Plan (fka ViacomCBS Excess 401(k) Plan) Paramount Global Excess 401(k) Plan for Designated Senior Executives (fka ViacomCBS Excess 401(k) Plan for Designated Senior Executives) Paramount Global Bonus Deferral Plan (fka ViacomCBS Bonus Deferral Plan) Paramount Global Bonus Deferral Plan for Designated Senior Executives (fka ViacomCBS Bonus Deferral Plan for Designated Senior Executives)

(Full title of the plan)

Christa A. D'Alimonte Executive Vice President, General Counsel and Secretary Paramount Global 1515 Broadway New York, New York 10036 (212) 258-6000 (Name, address, including zip code, and telephone number, including area code, of agent for service)

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act. Large accelerated filer Accelerated filer Accelerated filer

Large accelerated filer Non-accelerated filer Accelerated filer□Smaller reporting company□Emerging growth company□

If an emerging growth company, indicate by check mark whether the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 7(a)(2)(B) of the Securities Act.

#### EXPLANATORY NOTE

This Registration Statement has been filed for the purpose of registering an additional \$150,000,000 of unsecured obligations of the Company to pay deferred compensation in the future ("Deferred Compensation Obligations") in accordance with the terms of the Paramount Global Excess 401(k) Plan (fka ViacomCBS Excess 401(k) Plan), Paramount Global Excess 401(k) Plan for Designated Senior Executives), Paramount Global Bonus Deferral Plan (fka ViacomCBS Bonus Deferral Plan) and Paramount Global Bonus Deferral Plan for Designated Senior Executives) (fka ViacomCBS Bonus Deferral Plan for Designated Senior Executives (fka ViacomCBS Bonus Deferral Plan for Designated Senior Executives) (collectively, the "Plans"), which are in addition to the \$100,000,000 in Deferred Compensation Obligations registered on each of the registrant's Registration Statements on Form S-8 filed with the Commission on December 21, 2001 and December 11, 2006 (Reg. No. 333-75752 and Reg. No. 333-139248, respectively) and the \$200,000,000 in Deferred Compensation Statement on Form S-8 filed with the Commission on December 5, 2013 (Reg. No. 333-192673) (the "2013 Statement"). Pursuant to General Instruction E to Form S-8, the contents of the 2013 Statement are incorporated by reference herein and made a part of this Registration Statement, except as superseded by corresponding exhibits as presented below in Part II, Item 8. Exhibits.

#### PART II

#### INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

#### Item 8. Exhibits.

- 3.1 Amended and Restated Certificate of Incorporation of Paramount Global, effective as of February 16, 2022 (incorporated by reference to Exhibit 3.1 to Paramount Global's Current Report on Form 8-K filed on February 18, 2022)
- 3.2 Amended and Restated Bylaws of Paramount Global, effective as of February 16, 2022 (incorporated by reference to Exhibit 3.2 to Paramount Global's Current Report on Form 8-K filed on February 18, 2022)
- 4.1 Paramount Global Excess 401(k) Plan (fka ViacomCBS Excess 401(k) Plan)\*
- 4.2 <u>Amendment No. 1 to the Paramount Global Excess 401(k) Plan (fka ViacomCBS Excess 401(k) Plan)</u>\*
- <u>4.3</u> Paramount Global Excess 401(k) Plan for Designated Senior Executives (fka ViacomCBS Excess 401(k) Plan for Designated Senior Executives).
   <u>- Part A (as amended and restated as of October 1, 2021) (incorporated by reference to Exhibit 10(b) to the Quarterly Report on Form 10-Q of ViacomCBS Inc. for the quarter ended September 30, 2021) (File No. 001-09553)</u>
- <u>4.4</u> Paramount Global Excess 401(k) Plan for Designated Senior Executives (fka ViacomCBS Excess 401(k) Plan for Designated Senior Executives)
   <u>- Part B (as amended and restated as of October 1, 2021) (incorporated by reference to Exhibit 10(c) to the Quarterly Report on Form 10-Q of ViacomCBS Inc. for the guarter ended September 30, 2021) (File No. 001-09553)
  </u>
- 4.5 Amendment No. 1 to the Paramount Global Excess 401(k) Plan for Designated Senior Executives (fka ViacomCBS Excess 401(k) Plan for Designated Senior Executives) (as amended and restated as of October 1, 2021)\*
- 4.6 Paramount Global Bonus Deferral Plan (fka ViacomCBS Bonus Deferral Plan)\*
- 4.7 Amendment No. 1 to the Paramount Global Bonus Deferral Plan (fka ViacomCBS Bonus Deferral Plan)\*
- <u>4.8</u> <u>Paramount Global Bonus Deferral Plan for Designated Senior Executives (fka ViacomCBS Bonus Deferral Plan for Designated Senior Executives) Part A (as amended and restated as of October 1, 2021) (incorporated by reference to Exhibit 10(e) to the Quarterly Report on Form 10-Q of ViacomCBS Inc. for the quarter ended September 30, 2021) (File No. 001-09553)</u>
- <u>4.9</u> Paramount Global Bonus Deferral Plan for Designated Senior Executives (fka ViacomCBS Bonus Deferral Plan for Designated Senior Executives) Part B (as amended and restated as of October 1, 2021) (incorporated by reference to Exhibit 10(f) to the Quarterly Report on Form 10-Q of ViacomCBS Inc. for the quarter ended September 30, 2021) (File No. 001-09553)
- 4.10 Amendment No. 1 to the Paramount Global Bonus Deferral Plan for Designated Senior Executives (fka ViacomCBS Bonus Deferral Plan for Designated Senior Executives)\*
- 5.1 Opinion of Shearman & Sterling LLP\*
- 23.1 Consent of PricewaterhouseCoopers LLP, Independent Public Registered Accounting Firm for Paramount Global\*
- 23.2 Consent of Shearman & Sterling LLP (included in Exhibit 5.1) \*
- 24.1 Powers of Attorney\*
- <u>107</u> <u>Filing Fee Table</u>

\* Filed herewith

#### SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended (the "Securities Act"), the registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of New York, State of New York, on October 7, 2022.

# PARAMOUNT GLOBAL

By: /s/ Christa A. D'Alimonte

Name: Christa A. D'Alimonte Title: Executive Vice President, General Counsel and Secretary

Pursuant to the requirements of the Securities Act, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated.

<u>Signature</u>	<u>Title</u>	Date
Principal Executive Officer		
/s/ Robert M. Bakish	President, Chief Executive Officer and Director	October 7, 2022
Robert M. Bakish		
Principal Financial and Accounting Officers		
/s/ Naveen Chopra	Executive Vice President, Chief Financial Officer	October 7, 2022
Naveen Chopra		
/s/ Katherine M. Gill-Charest	Executive Vice President, Controller & Chief Accounting Officer	October 7, 2022
Katherine M. Gill-Charest		
Directors		
*	Director	October 7, 2022
Candace K. Beinecke		
*	Director	October 7, 2022
Barbara M. Byrne		
*	Director	October 7, 2022
Linda M. Griego		
*	Director	October 7, 2022
Robert N. Klieger		
*	Director	October 7, 2022
Judith A. McHale		
*	Director	October 7, 2022
Ronald L. Nelson		

	*	Director	October 7, 2022
Charles E. Phillips, Jr.		-	
Shari E. Redstone	*	Non-Executive Chair of the Board of Directors	October 7, 2022
Susan Schuman	*	Director	October 7, 2022
Nicole Seligman	*	Director	October 7, 2022
Frederick O. Terrell	*	Director	October 7, 2022

\*By: /s/ Christa A. D'Alimonte Christa A. D'Alimonte Attorney-in-Fact

# VIACOMCBS EXCESS 401(k) PLAN PART A

(2021 Restatement)

Section 1	1. Establishment and Purpose of the Plan	1
1.1	<u>Establishment</u>	1
1.2	Purpose	1
Section 2	2. <u>Definitions</u>	1
Section 3	3. <u>Participation</u>	3
3.1	Designation of Eligible Employees	3
3.2	Election to Participate	3
3.3	Amendment or Suspension of Election	4
3.4	Amount of Elections	4
Section 4	4. <u>Employer Match</u>	4
Section 5	5. <u>Individual Account</u>	5
5.1	Creation of Accounts	5
5.2	Joint Payment Account Option Election	5
5.3	Investments	6
5.4	Account Statements	7
Section 6	6. <u>Payment</u>	7
6.1	Payment on Account of Termination of Employment For Reasons Other Than Disability	7
6.2	Payment on Account of Disability	7
Section 7	7. <u>Nature of Interest of Participant</u>	8
Section 8	3. Hardship Distributions and Deferral Revocations	8
Section 9	9. <u>Beneficiary Designation</u>	8
Section 1	10. <u>Administration</u>	8
10.1	Administrative Committee	8
10.2	Powers of the Administrative Committee	8
10.3	<u>Claims Procedure</u>	9
10.4	Finality of Administrative Committee Determinations	9
10.5	<u>Severability</u>	9
10.7	Gender	9
	11. <u>No Employment Rights</u>	9
Section 1	12. Amendment, Suspension, and Termination	10

## Section 1. Establishment and Purpose of the Plan.

1.1 *Establishment.* The Viacom Excess 401(k) Plan was adopted as of April 1, 1984 as an unfunded plan of voluntarily deferred compensation for the benefit of Participants. As of December 31, 2005, it was renamed the CBS Excess 401(k) Plan, and as of October 1, 2021, it is hereby amended and restated as the ViacomCBS Excess 401(k) Plan. Any Eligible Employee who is identified by the Company on or after August 28, 2002 as a reporting person for purposes of Section 16 of the Securities Exchange Act of 1934 ("Reporting Employee") or any employee of an Employer who is eligible to participate in the Plan and whose securities may be attributable to a Reporting Employee for purposes of Section 16 of the Securities Exchange Act of 1934 shall no longer be eligible to participate in this Plan, and shall instead be eligible to participate in the CBS Excess 401(k) Plan for Designated Senior Executives, or its successors, including the ViacomCBS Excess 401(k) Plan for Designated Senior Executive Excess Plan"). Any deferrals made under the Plan by any Reporting Employee who was a participant in the Plan on August 28, 2002 and by any Reporting Employee (or any other Eligible Employee whose securities may be attributable to a Reporting Employee (or any other Eligible Employee whose securities may be attributable to a Reporting Employee (or any other Eligible Employee who was a participant in the Plan on August 28, 2002 and by any Reporting Employee (or any other Eligible Employee whose securities may be attributable to a Reporting Employee) prior to the date he becomes a Reporting Employee (or the date his securities are attributable to a Reporting Employee) were transferred to the Executive Excess Plan as of December 1, 2005 or, if later, as of the date he becomes a Reporting Employee).

1.2 *Purpose*. The purpose of Part A of this Plan is to provide the means by which an Eligible Employee could have, in certain circumstances, elected to defer receipt of a portion of his Compensation prior to January 1, 2005. On or after January 1, 2005, all compensation is deferred under Part B of this Plan. The Plan also provides that the Company could, in certain instances, credit the Account of a Participant with an Employer Match.

# Section 2. Definitions.

The following words and phrases as used in Part A of this Plan have the following meanings:

2.1 The term "Account" shall mean a Participant's individual account, as described in Section 5 of the Plan.

2.2 The term "Administrative Committee" means (i) for the periods prior to June 1, 2020, the CBS Retirement Committee, and (ii) for the periods on and after June 1, 2020, the ViacomCBS Administrative Committee. The Administrative Committee may act on its own behalf or through the actions of its duly authorized delegates.

2.3 The term "Board of Directors" means the Board of Directors of the Company.

2.4 The term "Bonus" means any cash bonus paid under the CBS Corporation Short-Term Incentive Plan and any other comparable annual cash bonus plan sponsored by any Employer.

2.5 The term "Company" shall refer to (i) CBS Corporation and its subsidiaries for periods prior to December 4, 2019, and (ii) ViacomCBS Inc. for the periods on and after December 4, 2019.

2.6 The term "Compensation" means an Eligible Employee's annual compensation as defined in the CBS 401(k) Plan (and its successor the ViacomCBS 401(k) Plan, as applicable), with the following modifications: (i) the limitations imposed by Internal Revenue Code §401 (a)(17) shall not be taken into account, and (ii) Bonuses earned for calendar years prior to January l, 2002 shall not be excluded.

2.7 The term "Disability" or "Disabled" means that a Participant (i) has been determined to be disabled by the Social Security Administration, or (ii) is receiving benefits under the provisions of the long-term disability plan covering such Participant that is sponsored by or participated in by the Participant's Employer. The date a Participant meets the definition of Disability shall be treated as the date he terminates employment for purposes of Section 6 of the Plan.

2.8 The term "Eligible Employee" means an employee of an Employer (i) for whom the sum of (a) the rate of annual base salary for a particular year and (b) actual commissions received for the prior year, equals or is greater than the annual compensation limit in effect under Internal Revenue Code Section 401(a)(17) (as adjusted from time to time by the Administrative Committee), and (ii) is designated by the Administrative Committee as an employee who is eligible to participate in the Plan. If an employee becomes an Eligible Employee in any Plan Year, such employee shall remain an Eligible Employee for all future Plan Years; provided, however, that the Administrative Committee may terminate such employee's eligibility for the Plan if his annual base salary as of January 1 of any Plan Year is less than the amount in clause (i) in effect for the Plan Year in which such employee initially became an Eligible Employee. In no event shall any Reporting Employee be considered an Eligible Employee under the Plan on or after August 28, 2002.

2.9 The term "Employer" means the Company and any affiliate or subsidiary that adopts the Plan on behalf of its Eligible Employees.

2.10 The term "Employer Match" means the amounts credited to a Participant's Account with respect to a Participant's Excess Salary Reduction Contributions and Excess Bonus Deferral Contributions, calculated using the rate of matching contributions under the CBS 401(k) Plan, in effect at the time such Plan contributions were made. Effective January 1, 2002 for all Bonuses earned for calendar years beginning after December 31, 2001, Excess Bonus Deferral Contributions shall not be credited with an Employer Match.

2.11 For all Bonuses earned for calendar years prior to January 1, 2002, the term "Excess Bonus Deferral Contributions" means the portion of the Participant's Compensation attributable to a Bonus that he elects to defer under the terms of this Plan. Effective August 28, 2002 for all Bonuses earned on or after January 1, 2002, the Plan shall no longer provide for Excess Bonus Deferral Contributions. Any Bonus Deferral Contribution election made under this Plan for the Bonus earned for the calendar year 2002 shall be deemed to have been made under, and be recognized by, the CBS Bonus Deferral Plan, or the CBS Bonus Deferral Plan for Designated Senior Executives (or their successors), as appropriate.

2.12 The term "Excess Salary Reduction Contributions" means the portion of a Participant's Compensation, excluding any Bonus, earned during a Plan Year (after such Participant has reached any Limitation) that he elects to defer under the terms of this Plan.

2.13 The term "Investments Committee" means (i) for the periods prior to June 1, 2020, the Investments Committee for CBS Defined Contribution Plans, and (ii) for the periods on and after June 1, 2020, the ViacomCBS Investments Committee.

2.14 The term "Investment Options" means (i) prior to January 1, 2014, the investment funds available to participants in the CBS 401(k) Plan, excluding the Self-Directed Brokerage Account, (ii) effective as of January 1, 2014 through December 31, 2014, the notional investment options selected by the CBS Retirement Committee in its sole discretion, and (iii) effective as of January 1, 2015 means the notional investment options elected by the Investments Committee.

2.15 The term "Joint Payment Option" means, in accordance with Section 5.2, (i) any payment option election made by a Participant in effect in this Plan immediately prior to August 28, 2002, and (ii) any payment option election made on or after August 28, 2002. A Joint Payment Option shall apply to all amounts credited to the Participant's Account in this Plan and his account in the CBS Bonus Deferral Plan, or as applicable its successors, including the ViacomCBS Bonus Deferral Plan, as well as any similar plan applicable to Reporting Employees.

2.16 The term "Limitation" means the limitation on contributions to defined contribution plans under Section 415(c), on compensation taken into account under Section 401(a)(17), or on elective deferrals under Section 401(k)(3) and Section 402(g) of the Internal Revenue Code of 1986.

2.17 The term "Participant" means an Eligible Employee who elects to have Excess Salary Reduction Contributions or Excess Bonus Deferral Contributions made to the Plan.

2.18 The term "Plan" means (i) for the periods prior to October 1, 2021, the CBS Excess 401(k) Plan, and (ii) for the periods on and after October 1, 2021, the ViacomCBS Excess 401(k) Plan.

#### Section 3. Participation.

3.1 *Designation of Eligible Employees.* All employees who were Eligible Employees immediately prior to August 28, 2002 remained Eligible Employees, subject to Section 2.8. Beginning August 28, 2002 until December 31, 2004, each month the Administrative Committee could designate in its sole discretion those additional employees who satisfy the terms of paragraph 2.7 as eligible to participate in the Plan. On and after January 1, 2005, all Compensation is deferred under Part B of the Plan.

3.2 *Election to Participate.* (a) Prior to January 1, 2005, Eligible Employees could elect to participate in Part A of the Plan. An Eligible Employee could elect, at any time after becoming eligible, to participate and make Excess Salary Reduction Contributions during the Plan Year by filing an election with the Administrative Committee in accordance with this Section 3 and the rules and regulations established by the Administrative Committee. Such election was effective on a prospective basis beginning with the payroll period that occurred as soon as was administratively practicable following receipt of the election by the Administrative Committee, until December 31, 2004.

(b) For Bonuses earned for calendar years prior to January l, 2002, an Eligible Employee could elect within 30 days of the date he became an Eligible Employee to make an Excess Bonus Deferral Contribution with respect to any Bonus scheduled to be paid in the next succeeding calendar year. Prior to December 31 of each Plan Year, an Eligible Employee could elect to make an Excess Bonus Deferral Contribution with respect to any Bonus scheduled to be paid in the second succeeding calendar year. For example, prior to December 31, 1999 an Eligible Employee could make an Excess Bonus Deferral Contribution election with respect to any cash bonus scheduled to be paid in 2001 under the CBS Corporation Short-Term Incentive Plan.

3.3 Amendment or Suspension of Election. Prior to January 1, 2005, Participants could change (including, suspend) their existing Excess Salary Reduction Contribution election under Part A of this Plan during the Plan Year by filing a new election in accordance with the prescribed administrative guidelines. Such new election was effective on a prospective basis beginning with the payroll period that occurred as soon as administratively practicable following receipt of the election by the Administrative Committee. A Participant was not permitted to make up suspended Excess Salary Reduction Contributions, and during any period in which a Participant's Excess Salary Reduction Contributions were suspended, the Employer Match to the Plan was suspended. Any election was irrevocable once made and is invalid if made beyond the dates prescribed in paragraph 3.2.

3.4 *Amount of Elections.* (a) Each election filed by an Eligible Employee must have specified the amount of Excess Salary Reduction Contributions in a whole percentage between 1% and 15% of the Participants' Compensation, excluding any Bonus.

(b) For all Bonuses earned for calendar years prior to January 1, 2002, each Bonus Deferral election filed by an Eligible Employee must have specified the amount of Excess Bonus Deferral Contribution in a whole percentage between 1% and 15% of the Participant's applicable Bonus.

(c) For Eligible Employees as of December 31, 1995, Compensation for Plan Year 1997 subject to Excess Salary Reduction Contributions and Excess Bonus Deferral Contributions did not exceed the greater of (i) \$750,000, or (ii) such Eligible Employee's compensation, as determined by the Administrative Committee, for the 1995 Plan Year. For employees who become Eligible Employees in 1996 or 1997, Compensation for Plan Years 1996 and 1997, if applicable, subject to Excess Salary Reduction Contributions and Excess Bonus Deferral Contributions did not exceed \$750,000.

#### Section 4. Employer Match.

Prior to January 1, 2005, an Employer Match was credited approximately every two weeks to a Participant's Account with respect to the eligible portion of Excess Salary Reduction Contributions and, for Bonuses earned for calendar years beginning prior to January 1, 2002, Excess Bonus Deferral Contributions, of such Participant. The eligible portion of a Participant's Excess Salary Reduction Contributions and the eligible portion of a Participant's Excess Bonus Deferral Contribution. For Eligible Employees as of December 31, 1995, the eligible portion of such Participant's Excess Bonus Deferral Contribution for the 1997 Plan Year and each subsequent year was based on Compensation up to an annual maximum equal to the greater of (i) \$750,000, or (ii) such Eligible Employee's compensation, as determined by the Administrative Committee, for the 1995 Plan Year. For employees who become eligible in 1996 and subsequent years, the eligible portion of a Participant's Excess Salary Reduction Contributions and the eligible portion of a Participant's Excess Bonus Deferral Committee, for the 1995 Plan Year. For employees who become eligible portion of a Participant's Excess Bonus Deferral Contributions and the eligible portion of a Participant's Excess Bonus Deferral Contributions and the eligible portion of a Participant's Excess Bonus Deferral Contributions and the eligible in 1996 and subsequent years, the eligible portion of a Participant's Excess Salary Reduction Contributions and the eligible portion of a Participant's Excess Bonus Deferral Contribution was based on Compensation up to an annual maximum amount of \$750,000. Notwithstanding the foregoing, for any Participant who was also a participant in the new Viacom 401(k) Plan and either the new Viacom Excess 401(k) Plan or the new Viacom Excess 401(k) Plan or the new Viacom Excess 401(k) Plan for Designated Senior Executives after December 31, 2005, the maximum amount of compensation with respect to which matching contributions were made was limited to \$375,000.

#### Section 5. Individual Account.

5.1 *Creation of Accounts.* The Company will maintain an Account in the name of each Participant. Each Participant's Account will be credited with the amount of the Participant's (i) Excess Salary Reduction Contributions earned for calendar years prior to January 1, 2005, (ii) Excess Bonus Deferral Contributions for Bonuses earned for calendar years prior to January 1, 2002, and (iii) Employer Match, if any, made in all Plan Years prior to January 1, 2005.

5.2 *Joint Payment Account Option Election.* (a) Any Joint Payment Option defined in Section 2.15(i) shall continue to apply until changed by the Participant in accordance with this Section 5.

(b) Any Eligible Employee who first becomes a Participant on or after August 28, 2002 and who has not elected a Joint Payment Option under Section 4.2 of the CBS Bonus Deferral Plan, or its successors, was required to elect a Joint Payment Option at the same time that the Participant filed his initial election to commence participation in the Plan pursuant to Section 3.2. Such Joint Payment Option shall continue to apply until changed by the Participant in accordance with this Section 5.

(c) A Participant may elect to receive his entire Account under either of the following Joint Payment Options: (i) a single lump sum; or, (ii) annual payments over a period of two, three, four or five years on or about January 31<sup>st</sup> beginning in the calendar year immediately following the end of the Plan Year in which the Participant terminates employment. If no Joint Payment Option election is made in accordance with the terms of the Plan or under the CBS Bonus Deferral Plan, or its successors, a Participant shall be deemed to have elected to receive his Account in a single lump sum on or about January 31<sup>st</sup> of the calendar year immediately following the end of the Plan Year in which the Participant terminates employment. If a Participant makes a Joint Payment Option election to receive payments in a single lump sum, such lump sum shall be payable on or about January 31<sup>st</sup> of the calendar year immediately following the end of the Plan Year in which the Participant terminates employment, unless the Participant elects to be paid on or about January 31<sup>st</sup> of the second, third, fourth, or fifth calendar year following the year in which the Participant terminates employment. If a Participant elects to receive annual payments over a period of two or more years, such annual payments shall be made in substantially equal annual payments, unless the Participant designates, at the time of making his Joint Payment Option election, a specific percentage of his Account to be distributed in each year. All specified percentages must be a whole multiple of 10% and the total of all designated percentages must be equal to 100%.

Example 1: If a Participant elects (or is deemed to elect) a Joint Payment Option that provides for a lump sum payment and terminates employment in 2022, such lump sum shall be paid on or about January 31<sup>st</sup>, 2023. A Participant alternatively could designate January 31<sup>st</sup> of 2024, 2025, 2026 or 2027 in which to receive his lump sum.

Example 2: If a Participant elects a Joint Payment Option that provides for annual payments over a period of four years and terminates employment in 2022, each payment on or about January 31<sup>st</sup>, 2023 through 2026 will be comprised of approximately 25% of the Participant's Account as of the Participant's date of termination. A Participant alternatively could designate 10% of his Account to be distributed in January 2023, 20% in January 2024, 30% in January 2025 and 40% in January 2026; or, any other combination of percentages that totals 100%.

(d) A Participant may change his Joint Payment Option no more than three times over the course of his employment with the Company or any affiliate. A Participant may change an existing Joint Payment Option only one time in any calendar year. Any change of a Participant's existing Joint Payment Option election made less than six months prior to the Participant's termination of employment for any reason shall be null and void, and the Participant's last valid Joint Payment Option shall remain in effect.

5.3 *Investments.* (a) Prior to January 1, 2014, all Excess Salary Reduction Contributions, Excess Bonus Deferral Contributions and Employer Match, if any, were credited through December 31<sup>st</sup> of the calendar year in which the Participant terminates employment with an amount equal to such amount which would have been earned had such contributions been invested in the same Investment Options and in the same proportion as the Participant may elect, from time to time, to have his Salary Reduction Contributions and Matching Employer Contributions invested under the CBS 401(k) Plan; or if no such election has been made, in the PRIMCO Stable Value Fund (or any successor fund). Effective February 1, 2015, at the time an employee becomes a Participant under the Plan, and for existing Participants, prior to the time of the first payroll period occurring on or after February 1, 2015, each Participant shall elect, in a manner determined by the Administrative Committee, that his future Matching Employer contributions, if any, be notionally invested in multiples of 1% in any one or more of the Investment Options available under the Plan. Each Participant who does not, as of such time, make such an investment election, shall have his future Matching Employer Contributions notionally invested in such Investment Option(s) as may be selected by the Administrative Committee. (which Investment Option(s) shall be a "qualified default investment alternative" within the meaning of Department of Labor regulations.)

Effective as of January 1, 2014, a Participant may select from a list of notional Investment Options how the balance of his or her Account will be invested. If no selection is made, the Participant's Account will be notionally invested in the "qualified default investment alternative" (within the meaning of the CBS 401(k) Plan, or its successors, including the ViacomCBS 401(k) Plan) in effect from time to time. Earnings and losses received on the Participant's notional investments will be credited to the Participant's Account in the manner designated by the Investments Committee. The Investments Committee shall develop such procedures as it, in its discretion, deems advisable with respect to the selection of notional investments by Participants and the reflection of value attributable to such notional investments in their Accounts, including, without limitation, procedures which restrict a Participant's ability to notionally invest in certain Investment Options.

(b) (i) Prior to October 2, 2017, if a terminated Participant elected (or was deemed to elect) a single lump sum Joint Payment Option payable in the first calendar year following the calendar year in which the Participant terminated employment, no additional adjustments were made to the Participant's Account after December 31<sup>st</sup> of the calendar year in which the Participant terminated employment. If a Participant elected a single lump sum Joint Payment Option payable in the second, third, fourth or fifth calendar year following the calendar year in which the Participant terminated employment, the Participant's Account was credited with earnings based on the rate of return in the PRIMCO Stable Value Fund (or any successor fund) beginning January 1<sup>st</sup> of the calendar year inmediately preceding the calendar year in which the single lump sum was paid.

(ii) Effective October 2, 2017, if a terminated Participant elected (or is deemed to elect) a single lump sum Joint Payment Option payable in the first calendar year following the calendar year in which the Participant terminates employment, or if the Participant elects a single lump sum Joint Payment Option payable in the second, third, fourth or fifth calendar year following the calendar year in which the Participant terminates employment, the Participant's Account shall be credited with earnings based on the rate of return of his selected Investment Options (or such Investment Options selected by the Investments Committee) until such time as the date upon which the single lump sum payment is determined and paid.

(c) (i) Prior to October 2, 2017, if a terminated Participant elected annual payments, no additional adjustments were made to any amount payable in the first calendar year following the year in which the Participant terminated employment. For any annual payments made in the second, third, fourth or fifth year following the calendar year in which the Participant terminated employment, the Participant's Account was credited with earnings based on the rate of return in the PRIMCO Stable Value Fund (or any successor fund) beginning January 1<sup>st</sup> of the calendar year following the year in which the Participant terminated employment and continuing through December 31<sup>st</sup> of the calendar year immediately preceding the calendar year in which each payment was made.

(ii) Effective October 2, 2017, if a terminated Participant elects annual payments, the Participant's Account shall be credited with earnings based on the rate of return in his selected Investment Options (or such Investment Options selected by the Investments Committee) until such time as the date upon which each annual payment is determined and paid.

(d) No provision of this Plan shall require the Company or the Employer to actually invest any amounts in any fund or in any other investment vehicle.

5.4 *Account Statements*. Each Participant will be given, at least annually, a statement showing (i) the amount of all Contributions, (ii) the amount of Employer Match, if any, made with respect to his Account for such Plan Year, and (iii) the balance of the Participant's Account after crediting Investments.

#### Section 6. Payment.

6.1 *Payment on Account of Termination of Employment For Reasons Other Than Disability.* A Participant (or a Participant's beneficiary) shall be paid the balance in his Account following termination of employment in accordance with the Joint Payment Option in effect with respect to the Participant.

6.2 Payment on Account of Disability. A Participant (or a Participant's beneficiary) shall be paid the balance in his Account following the date he meets the definition of Disability in accordance with the Joint Payment Option in effect with respect to the Participant. If a Participant no longer meets the definition of Disability and returns to work with an Employer, no further payments shall be made on account of the prior Disability, and distribution of his remaining Account shall be made as otherwise provided in this Section 6 at the time of his subsequent termination of employment.

#### Section 7. Nature of Interest of Participant.

Participation in this Plan will not create, in favor of any Participant, any right or lien in or against any of the assets of the Company or any Employer, and all amounts of Compensation deferred hereunder shall at all times remain an unrestricted asset of the Company or the Employer. A Participant's rights to benefits payable under the Plan are not subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, or encumbrance. All payments hereunder shall be paid in cash from the general funds of the Company or applicable Employer and no special or separate fund shall be established and no other segregation of assets shall be made to assure the payment of benefits hereunder. Nothing contained in this Plan, and no action taken pursuant to its provisions, shall create or be construed to create a trust of any kind, or a fiduciary relationship, between any Employer and a Participant or any other person, and the Company's and each Employer's promise to pay benefits hereunder shall at all times remain unfunded as to the Participant.

#### Section 8. Hardship Distributions and Deferral Revocations.

A Participant may request the Administrative Committee to accelerate distribution of all or any part of the value of his Account solely for the purpose of alleviating an immediate financial emergency. For purposes of the Plan, such an immediate financial emergency shall mean an unanticipated emergency that is caused by an event beyond the control of the Participant and which would result in severe financial hardship to the Participant if early distribution were not permitted. The Administrative Committee may request that the Participant provide certifications and other evidence of qualification for such emergency hardship distribution as it determines appropriate. The decision of the Administrative Committee with respect to the grant or denial of all or any part of such request shall be in the sole discretion of the Administrative Committee, whether or not the Participant demonstrates an immediate financial emergency exists, and shall be final and binding and not subject to review.

#### Section 9. Beneficiary Designation.

A Participant's beneficiary designation for this Plan will automatically be the same as the Participant's beneficiary designation recognized under the CBS 401(k) Plan, or its successors including the ViacomCBS 401(k) Plan, unless a separate Designation of Beneficiary Form for this Plan has been properly filed.

#### Section 10. Administration.

10.1 *Administrative Committee*. This Plan will be administered by the Administrative Committee, the members of which will be selected by the Board of Directors.

10.2 *Powers of the Administrative Committee*. The Administrative Committee's powers will include, but will not be limited to, the power:

(i) to determine who are Eligible Employees for purposes of participation in the Plan;

(ii) to interpret the terms and provisions of the Plan and to determine any and all questions arising under the Plan, including without limitation, the right to remedy possible ambiguities, inconsistencies, or omissions by a general rule or particular decision;

- (iii) to adopt rules consistent with the Plan; and
- (iv) to approve certain amendments to the Plan.

10.3 Claims Procedure. The Administrative Committee shall have the exclusive right to interpret the Plan and to decide any and all matters arising thereunder. In the event of a claim by a Participant as to the amount of any distribution or method of payment under the Plan, within 90 days of the filing of such claim, unless special circumstances require an extension of such period, such person will be given notice in writing of any denial, which notice will set forth the reason for the denial, the Plan provisions on which the denial is based, an explanation of what other material or information, if any, is needed to perfect the claim, and an explanation of the claims review procedure. The Participant may request a review of such denial within 60 days of the date of receipt of such denial by filing notice in writing with the Administrative Committee. The Participant will have the right to review pertinent Plan documents and to submit issues and comments in writing. The Administrative Committee will respond in writing to a request for review within 60 days of receiving it, unless special circumstances require an extension of such period. The Administrative Committee, at its discretion, may request a meeting to clarify any matters deemed appropriate.

10.4 *Finality of Administrative Committee Determinations*. Determinations by the Administrative Committee and any interpretation, rule, or decision adopted by the Administrative Committee under the Plan or in carrying out or administering the Plan shall be final and binding for all purposes and upon all interested persons, their heirs, and personal representatives.

10.5 *Severability*. If a provision of the Plan shall be held illegal or invalid, the illegality or invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included in the Plan.

10.6 *Governing Law.* The provisions of the Plan shall be governed by and construed in accordance with the laws of the State of New York, to the extent not preempted by the laws of the United States.

10.7 *Gender*. Wherein used herein, words in the masculine form shall be deemed to refer to females as well as males.

#### Section 11. No Employment Rights.

No provisions of the Plan or any action taken by the Company, the Board of Directors, or the Administrative Committee shall give any person any right to be retained in the employ of any Employer, and the right and power of the Company to dismiss or discharge any Participant is specifically reserved.

## Section 12. Amendment, Suspension, and Termination.

The Administrative Committee shall have the right to amend the Plan at any time, unless provided otherwise in the Company's governing documents. The Board of Directors shall have the right to suspend or terminate the Plan at any time. No amendment, suspension or termination shall, without the consent of a Participant, adversely affect such Participant's rights in his account. In the event the Plan is terminated, the Administrative Committee shall continue to administer the Plan in accordance with the relevant provisions thereof.

IN WITNESS WHEREOF, in accordance with the Administrative Committee's August 20, 2021 Unanimous Written Consent, the Company has caused this Plan to be executed by its duly authorized representative this 20th day of September, 2021.

ViacomCBS Inc.

By:	/s/ Mark Beatty	
	Mark Beatty	

Its: Member, ViacomCBS Administrative Committee

VIACOMCBS EXCESS 401(k) PLAN PART B

# (2021 Restatement)

Section	1. Establishment and Purpose of the Plan	1
1.1	<u>Establishment</u>	1
1.2	Amendment and Restatement	1
1.3	Merger	1
1.4	Reporting Employees	1
1.5	Purpose	1
Section	2. <u>Definitions</u>	2
Section	3. <u>Participation</u>	5
3.1	Designation of Eligible Employees	5
3.2	Election to Participate	6
3.4	Amount of Elections	6
Section	4. Employer Match	7
Section	5. <u>Vesting</u>	7
Section	6. Individual Account	8
6.1	Creation of Accounts	8
6.2	Investments	8
6.3	Account Statements	10
Section	7. <u>Payment</u>	10
7.1	Joint Payment Option Election	10
7.2	Payment on Account of Separation from Service	12
7.3	Payment on Account of Participant's Death	12
Section	8. Unforeseeable Emergency Distributions and Deferral Revocations	12
Section	9. <u>Beneficiary Designation</u>	12
Section	10. Nature of Interest of Participant	13
Section	11. <u>Administration</u>	13
11.1	Committee	13
11.2	Powers of the Committee	13
11.3	Claims Procedure	14
11.4	Finality of Committee Determinations and Delegation	15
11.5	Rules and Regulations Established by Committee	15
Section	12. <u>No Employment Rights</u>	16
Section	13. Amendment, Suspension, and Termination	16
Section	14. <u>Miscellaneous</u>	16
14.1	<u>Severability</u>	16
14.2	Governing Law	16
14.3	Gender	16
14.4	Code Section 409A	16

#### Section 1. Establishment and Purpose of the Plan.

1.1 *Establishment*. The Viacom Excess 401(k) Plan was adopted as of April 1, 1984 as an unfunded plan of voluntarily deferred compensation for the benefit of Participants. On December 31, 2005, it was amended and restated and renamed the CBS Excess 401(k) Plan, and as of October 1, 2021, it was amended and restated and renamed the ViacomCBS Excess 401(k) Plan.

1.2 Amendment and Restatement. The Plan was again amended and restated effective as of January 1, 2009 by the adoption of Part B of the Plan, as set forth herein, and is hereby amended and restated effective as of October 1, 2021. Part A of the Plan, consisting of the original Plan adopted April 1, 1984, along with certain amendments, applies to compensation that was Deferred prior to January 1, 2005 in accordance with the terms of those documents in effect from time to time prior to October 3, 2004, subject to certain exceptions set forth in Part A of the Plan. The provisions of this Part B apply to compensation that is Deferred on or after January 1, 2005. This Part B of the Plan is intended to meet all of the requirements of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code"), so that Participants will be eligible to defer the receipt of, and the liability for the federal income tax with respect to, certain items of compensation from one year to a later year in accordance with the provisions of the Plan.

1.3 *Merger*. Under an agreement and plan of merger between CBS Corporation and Viacom Inc., dated August 13, 2019, Viacom Inc. merged with and into CBS Corporation on December 4, 2019, to become ViacomCBS Inc.

1.4 *Reporting Employees*. Any Eligible Employee who is identified by the Company on or after August 28, 2002 as a reporting person for purposes of Section 16 of the Securities Exchange Act of 1934 ("Reporting Employee") or any employee of an Employer who is eligible to participate in the Plan and whose securities may be attributable to a Reporting Employee for purposes of Section 16 of the Securities Exchange Act of 1934 shall no longer be eligible to participate in the Plan, and shall instead be eligible to participate in the CBS Excess 401(k) Plan for Designated Senior Executives, or, as of October 1, 2021, its successor the ViacomCBS Excess 401(k) Plan for Designated Senior Executive Excess Plan"), effective as of the date on which he becomes a Reporting Employee (or the date his securities are attributable to a Reporting Employee). Any deferrals made under the Plan by any Reporting Employee whose securities may be attributable to a Reporting Employee) prior to the date he becomes a Reporting Employee (or the date his securities are attributable to a Reporting Employee) prior to the date he becomes a Reporting Employee (or the date his securities are attributable to a Reporting Employee) prior to the date he securities are attributable to a Reporting Employee) or to the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employee) or to the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employee (or the date his securities are attributable to a Reporting Employ

1.5 *Purpose*. The purpose of Part B of the Plan is to provide the means by which a select group of Eligible Employees may, in certain circumstances, elect to defer receipt of a portion of their Compensation received on or after January 1, 2005. The Plan also provides that the Company will, in certain instances, credit the Account of a Participant with an Employer Match.

#### Section 2. Definitions.

The following words and phrases as used in Part B of the Plan have the following meanings:

2.1 The term "Account" shall mean a Participant's individual account, as described in Section 6.1 of the Plan.

2.2 The term "Administrative Committee" means (i) for the periods prior to June 1, 2020, the CBS Retirement Committee, and (ii) for the periods on and after June 1, 2020, the ViacomCBS Administrative Committee. The Administrative Committee may act on its own behalf or through the actions of its duly authorized delegates.

2.3 The term "Annual Payments" is defined in Section 7.1(b)(i).

2.4 The term "Board of Directors" means the Board of Directors of the Company.

2.5 The term "Bonus" means any cash bonus paid under the STIP and any other annual cash bonus plan (or annual component of a cash bonus plan) sponsored by an Employer which, in the discretion of the Administrative Committee, is comparable to the STIP.

2.7 The term "Code" means the Internal Revenue Code of 1986, as amended.

2.8 The term "Company" shall refer to (i) CBS Corporation and its subsidiaries for periods prior to December 4, 2019, and (ii) ViacomCBS Inc. for the periods on and after December 4, 2019.

2.9 The term "Compensation" means an Eligible Employee's base pay for services rendered to an Employer paid during such Employer's payroll period, including all elective contributions made on behalf of an Eligible Employee either to a "qualified cash or deferred arrangement" (as defined under Code Section 401(k) and applicable regulations), a "cafeteria plan" (as defined under Code Section 125 and applicable regulations), or a "qualified transportation fringe" (as defined under Code Section 132(t) and applicable regulations) maintained by an Employer, plus all overtime pay, but excluding (x) deferred compensation, (y) additional compensation of every other kind, including cash bonuses under the Company's long-term incentive plans, and (z) any Bonus. Compensation shall be determined without taking into account the limitations imposed by Code section 401(a)(17). Notwithstanding the foregoing,

(i) With respect to any Eligible Employee who is an employee of Paramount Pictures Corporation and who is characterized as a "Production Auditor," Compensation shall include salary or wages paid to such Eligible Employee that are characterized by Paramount Pictures Corporation as "Idle Day Earnings"; and

(ii) With respect to any Eligible Employee who is characterized as "Talent" by his Employer, Compensation shall instead mean such Eligible Employee's talent benefit base, as determined under applicable policies of the Employer.

2.10 The term "Deferral Election" is defined in Section 3.2(a).

2.11 The term "Deferred" means that an amount is considered to be deferred within the meaning of Treasury Regulations sections 1.409A-6(a)(2) and 1.409A-6(a)(3).

2.12 The term "Disability" or "Disabled" means that a Participant (i) has been determined to be disabled by the Social Security Administration, or (ii) is receiving benefits under the provisions of the long-term disability plan covering such Participant that is sponsored by or participated in by the Participant's Employer.

2.13 The term "Election Agreement" is defined in Section 3.2(c).

2.14 The term "Election Filing Date" means, except as provided in Section 3.2(b), the date not later than the December 31 immediately preceding the first day of the applicable calendar year for which a particular Deferral Election is made.

2.15 The term "Eligible Employee" means an employee of an Employer (i) for whom the sum of (a) the rate of annual base salary for a particular year and (b) actual commissions received for the prior year, equals or is greater than the annual compensation limit in effect under Code Section 401 (as adjusted from time to time by the Administrative Committee); (ii) who is designated by the Administrative Committee as an employee who is eligible to participate in the Plan; and (iii) who is notified in writing (including by email or other electronic means) by the Administrative Committee that he is eligible to participate in the Plan. If an employee becomes an Eligible Employee with respect to any calendar year, such employee shall remain an Eligible Employee for all future calendar years; provided, however, that the Administrative Committee may terminate such employee's eligibility for the Plan with respect to a calendar year if his annual base salary as of January 1 of such calendar year is anticipated to be less than the amount in clause (i) in effect for the calendar year in which such employee under the Plan on or after August 28, 2002. With respect to any employee of an Employer who is characterized as "Talent" by his Employer, the requirement clause (i) above shall be met if the employee's talent benefit base, as determined under applicable policies of the Employer, equals or is greater than the annual compensation limit in effect under Code section 401(a)(17) (as adjusted from time to time by the Administrative Committee).

2.16 The term "Employer" means the Company and any affiliate or subsidiary that adopts the Plan on behalf of its Eligible Employees, except as provided in Section 2.31.

2.17 The term "Employer Match" means the amounts credited to a Participant's Account with respect to the Participant's Excess Salary Reduction Contributions, calculated using the rate of matching contributions under the CBS 401(k) plan, or its successors, including the ViacomCBS 401(k) Plan, in effect for the period to which such Plan contributions relate.

2.18 The term "Executive Excess Plan" is defined in Section 1.4.

2.19 The term "Excess Salary Reduction Contributions" means the portion of a Participant's Compensation that is earned during a calendar year after such Participant has reached any Limitation and that he elects to defer under the terms of the Plan.

2.20 The term "Investments Committee" means (i) for the periods prior to June 1, 2020, the Investments Committee for CBS Defined Contribution Plans, and (ii) for the periods on and after June 1, 2020, the ViacomCBS Investments Committee.

2.21 Prior to January 1, 2014, the term "Investment Options" means the investment funds available to participants in the CBS 401(k) Plan, excluding the Self-Directed Brokerage Account and effective as of January 1, 2014 through December 31, 2014, means the notional investment options selected by the CBS Retirement Committee in its sole discretion and effective as of January 1, 2015, means the notional investment options elected by the Investments Committee, in its sole discretion.

2.22 The term "Joint Payment Option" means the time and form of payment options available for the payment of an Account as described in Section 7.1.

2.23 The term "Joint Payment Option Election" is defined in Section 7.1.

2.24 The term "Limitation" means the limitation on (i) contributions to defined contribution plans under Code Section 415(c), (ii) compensation taken into account under Code Section 401(a)(17) or (iii) elective deferrals under Code Section 401(k) (3) and Code Section 402(g).

2.25 The term "Participant" means an Eligible Employee who elects to have Excess Salary Reduction Contributions made to the Plan.

2.26 The term "Payment Election" is defined in Section 7.1 (a).

2.27 The term "Plan" means (i) for the periods prior to October 1, 2021, the CBS Excess 401(k) Plan, and (ii) for the periods on and after October 1, 2021, the ViacomCBS Excess 401(k) Plan. Part A of the Plan is attached hereto and shall apply to Compensation which was Deferred prior to January 1, 2005. Part B of the Plan is set forth herein and shall apply to Compensation which is Deferred on or after January 1, 2005. Certain provisions of this Part B apply as of certain earlier effective dates as specified herein. References to "the Plan" shall be considered references to Part A and/or Part B of the Plan as context requires.

2.28 The term "Post-2004 Subaccount" is defined in Section 6.1.

2.29 The term "Pre-2005 Subaccount" is defined in Section 6.1.

2.30 The term "Reporting Employee" is defined in Section 1.4.

The term "Separation from Service" means the condition that exists when a Participant and the Employer 2.31 reasonably anticipate that no further services will be performed after a certain date or that the level of bona fide services that the Participant will perform after such date (whether as an employee or an independent contractor) would permanently decrease to no more than 20% of the average level of bona fide services performed (whether as an employee or an independent contractor) over the immediately preceding 36-month period (or the full period of services to the Employer if the Participant has been providing services to the Employer for less than 36 months). For purposes of this Section 2.31, for periods during which a Participant is on a paid bona fide leave of absence and has not otherwise experienced a Separation from Service, the Employee is treated as providing bona fide services at the level equal to the level of services that the Participant would have been required to perform to receive the compensation paid with respect to such leave of absence. Periods during which a Participant is on an unpaid bona fide leave of absence and has not otherwise experienced a Separation from Service are disregarded for purposes of this Section 2.31 (including for purposes of determining the applicable 36-month (or shorter) period). For purposes of this Section 2.31, and notwithstanding Section 2.16, the "Employer" shall be considered to include all members of the controlled group of corporations, trades or businesses which includes the Company; provided, however, that in applying Code Section 414(b), the phrase "at least 50 percent" shall be substituted for "at least 80 percent"; and in applying Code Section 414(c), the phrase "at least 50 percent" shall be used instead of the phrase "at least 80 percent." Separation from Service shall be determined on the basis of the modifications described in Treasury Regulation Section 1.409A-1(h)(3) (or any successor regulation)) as defined in Code Section 409A and the regulations or other guidance issued thereunder.

2.31 The term "STIP" means (i) for the periods prior to October 1, 2021, the CBS Short-Term Incentive Plan, and (ii) for the periods on and after October 1, 2021, the ViacomCBS Short-Term Incentive Plan.

2.32 The term "Unforeseeable Emergency" means an event that results in severe financial hardship to a Participant resulting from (a) an illness or accident of the Participant or his or her spouse, dependent (as defined in Code Section 152, without regard to Code Sections 152(b)(1), (b)(2), and (d)(1)(B)), or beneficiary, (b) loss of the Participant's property due to casualty, or (c) other similar extraordinary circumstances arising due to results beyond the control of the Participant. This Section 2.32 shall be interpreted in a manner consistent with Code Section 409A and applicable provisions of the Treasury Regulations.

2.33 The term "ViacomCBS 401(k) Plan" means the ViacomCBS 401(k) Plan (formerly known as the CBS 401(k) Plan), originally effective as of September 1, 2001, and as amended from time to time thereafter (or any successor plan).

## Section 3. Participation.

3.1 *Designation of Eligible Employees*. All employees who were Eligible Employees immediately prior to January 1, 2009 remained Eligible Employees, subject to Section 2.15. Beginning January 1, 2009, each month the Administrative Committee or its delegee will designate in its sole discretion those additional employees who satisfy the terms of Section 2.15 as eligible to participate in the Plan.

3.2 *Election to Participate*. (a) To participate in the Plan for a calendar year, an Eligible Employee must make an annual election (a "Deferral Election") to defer receipt of a specified portion of his or her Compensation for services rendered during such calendar year ("Excess Salary Reduction Contributions") in accordance with this Section 3. Subject to Section 3.2(b), such Deferral Election must be made not later than the Election Filing Date and shall be effective as of the Election Filing Date. An Eligible Employee's entitlement to make Excess Salary Reduction Contributions shall cease with respect to the calendar year following the calendar year in which he or she ceases to be an Eligible Employee.

(b) Notwithstanding the foregoing, an employee who first becomes an Eligible Employee during the course of a calendar year beginning on or after January 1, 2005 shall make a Deferral Election within 30 days following the date the employee first becomes an Eligible Employee, provided that such employee has not already become eligible to participate in any other account balance plan of the Employer (as modified by Section 2.31) that is required to be aggregated with the Plan under Code Section 409A. Such Deferral Election shall be effective on the date made and shall be effective with regard to Compensation earned during the portion of the calendar year following the filing of the Deferral Election with the Administrative Committee, as determined pursuant to the pro-ration method permitted under Code Section 409A. If an Eligible Employee is a participant in another account balance plan that is required to be aggregated with the Plan under Code section 409A when he first becomes eligible to participate in the Plan, such Eligible Employee shall be eligible to make a Deferral Election for the calendar year immediately following the calendar year of his initial eligibility by making an election in accordance with Section 3.2(a) above.

(c) All Deferral Elections shall be made on a written or electronic form acceptable to the Administrative Committee (an "Election Agreement") filed with the Administrative Committee and shall specify the percentage of a Participant's Compensation that is to be deferred under the Plan during the applicable calendar year.

(d) All Deferral Elections relating to calendar years beginning on or after January 1, 2005, once effective, shall be irrevocable for that calendar year. All Participants are required to make a Deferral Election for each calendar year. If an Eligible Employee fails to make a Deferral Election for a given calendar year, the Eligible Employee shall not be entitled to participate in the Plan during that calendar year. Such Eligible Employee may resume participation in the Plan by completing and filing with the Administrative Committee a new Deferral Election by the Election Filing Date for the succeeding calendar year(s).

3.3 *Amount of Elections*. Each Deferral Election filed by an Eligible Employee must specify the amount of Excess Salary Reduction Contributions in a whole percentage between 1% and 15% of the Eligible Employee's Compensation.

#### Section 4. Employer Match.

An Employer Match calculated using the same performance based formula that is used to credit matching contributions under (i) for the periods prior to October 1, 2021, the CBS 401(k) Plan, and (ii) for the periods on or after October 1, 2021, the ViacomCBS 401(k) Plan, will be credited each payroll period to a Participant's Account with respect to the eligible portion of Excess Salary Reduction Contributions to which an Employer Match has not previously been credited. For this purpose, the eligible portion of a Participant's Excess Salary Reduction Contributions shall be limited to 5% of such Excess Salary Reduction Contributions. For Participants who were Eligible Employees as of December 31, 1995, the eligible portion of such Participant's Excess Salary Reduction Contributions, as determined by the Administrative Committee, for the 1995 calendar year. For Participants who first became Eligible Employees in 1996 and subsequent calendar years, the eligible portion of such Participant's Excess Salary Reduction Contributions shall be based on Compensation up to an annual maximum amount of \$750,000. Notwithstanding the foregoing, for any Participant who is also a participant in the new Viacom 401(k) Plan and either the new Viacom Excess 401(k) Plan or the new Viacom Excess 401(k) Plan for Designated Senior Executives after December 31, 2005, the maximum amount of Compensation with respect to which an Employer Match will be made is limited to \$375,000.

#### Section 5. Vesting.

A Participant shall always be 100% vested in amounts credited to his Account hereunder, other than amounts attributable to an Employer Match.

Prior to July 1, 2019, a Participant's Employer Match became vested according to the following schedule:

Years of Vesting Service	Vesting %
Less than 1 year	0%
1 year but less than 2 years	20%
2 years but less than 3 years	40%
3 years but less than 4 years	60%
4 years but less than 5 years	80%
5 years or more	100%

Effective July 1, 2019 until December 31, 2020, a Participant's Employer Match will become vested according to the following schedule:

Years of Vesting Service	Vesting %
Less than 1 year	0%
1 year but less than 2 years	33 1/3%
2 years but less than 3 years	66 2/3%
3 years or more	100%

Effective as of January 1, 2021, a Participant will become vested in the Employer Match in accordance with the following schedule:

Years of Vesting Service	Vesting %
Less than 2 years	0%
2 years or more	100%

Notwithstanding the foregoing, a Participant will become fully vested in the Employer Match if such Participant attains age 65 or incurs a Disability while actively employed or terminates employment due to normal, early, or postponed retirement (determined under the terms of any tax qualified defined benefit plan maintained by the Employer), death, or Disability.

# Section 6. Individual Account.

6.1 *Creation of Accounts.* The Company will establish and maintain on its books a reserve Account in the name of each Participant. Each Participant's Account will be credited with the amount of the Participant's Excess Salary Reduction Contributions (and earnings and losses thereon) and Employer Match (and earnings and losses thereon), if any, made in all calendar years. A Participant's Account will be divided into the following subaccounts: (i) a "Pre-2005 Subaccount" for amounts Deferred as of December 31, 2004 (and earnings and losses thereon), and (ii) a "Post-2004 Subaccount" for amounts Deferred after December 31, 2004 (and earnings and losses thereon). Amounts in the Pre-2005 Subaccounts, which are intended to qualify for "grandfathered" status, shall be subject to the terms and conditions specified in Part A of the Plan as in effect prior to January 1, 2005.

6.2 *Investments*. (a) Prior to January 1, 2014, amounts, if any, in a Participant's Post-2004 Subaccount were credited through December 31<sup>st</sup> of the calendar year in which the Participant experiences a Separation from Service with an amount equal to the amount which would have been earned had such amounts been invested in the same Investment Options and in the same proportion as the Participant may elect, from time to time, to have his contributions invested under (i) for the periods prior to October 1, 2021, the CBS 401(k) Plan, and (ii) for the periods on or after October 1, 2021, the ViacomCBS 401(k) Plan (other than the Self Directed Account).

Effective as of January 1, 2014, a Participant may select from a list of notional Investment Options how the balance of his or her Account will be invested. If no selection is made, the Participant's Account will be notionally invested in the "qualified default investment alternative" within the meaning of (i) for the periods prior to October 1, 2021, the CBS 401(k) Plan, and (ii) for the periods on or after October 1, 2021, the ViacomCBS 401(k) Plan. Earnings and losses received on the Participant's notional investments will be credited to the Participant's Account in such a manner designated by the Investments Committee. The Investments Committee shall develop such procedures as it, in its discretion, deems advisable with respect to the selection of notional investments by Participants and the reflection of value attributable to such notional investments in their Accounts, including, without limitation, procedures which restrict a Participant's ability to notionally invest in certain Investment Options. Effective February 1, 2015, at the time an employee becomes a Participant under the Plan, and for existing Participants, prior to the time of the first payroll period occurring on or after February 1, 2015, each Participants shall elect, in a manner determined by the Investments Committee, that his future Matching Employer Contributions, if any, be notionally invested in multiples of 1% in any one or more of the Investment Options Available under the Plan. Each Participant who does not, as of such time, make such an investment election, shall have his future Matching Employer Contributions notionally invested in such Investment Option(s) as may be selected by the Investments Committee (or its predecessors, as applicable) of the Company (which Investment Option(s) shall be a "qualified default investment alternative" within the meaning of Department of Labor regulations).

(b) (i) Prior to October 2, 2017, when a Participant experienced a Separation from Service and elected (or was deemed to elect) to have his Post-2004 Subaccount distributed in a single lump sum, the Participant's Post 2004 Subaccount was credited with earnings based on the rate of return in the Fixed Income Fund (or any successor fund) beginning January 1<sup>st</sup> of the calendar year following the calendar year in which the Participant experienced a Separation from Service that resulted in the Participant's Post-2004 Subaccount becoming payable, and continuing through the date upon which such single lump sum payment was determined if such determination date was after December 31<sup>st</sup> of the calendar year in which the Participant experienced a Separation from Service. Payments due on January 31<sup>st</sup> of a calendar year were determined on the previous December 31<sup>st</sup>, while payments due on the first business day of a calendar month were determined on the last day of the second preceding calendar month (e.g., a payment scheduled for the first business day of March will be determined on the preceding January 31<sup>st</sup>).

(ii) Effective on and after October 2, 2017, a Participant who experienced or experiences a Separation from Service and elected (or is deemed to elect) to have his Post-2004 Subaccount distributed in a single lump shall have his Post-2004 Subaccount credited with earnings based on the rate of return in his selected Investment Options (or such Investment Options selected by the Investments Committee) until such time as the date upon which the single lump sum payment is determined and paid.

(c) (i) Prior to October 2, 2017, if a Participant experienced a Separation from Service and elected to have his Post-2004 Subaccount distributed in Annual Payments, the Participant's Post-2004 Subaccount were credited with earnings based on the rate of return in the Fixed Income Fund (or any successor fund) beginning January 1<sup>st</sup> of the calendar year following the calendar year in which the Participant experienced a Separation from Service that resulted in the Participant's Post-2004 Subaccount becoming payable, and continuing through the date upon which such Annual Payment was determined, if such determination date was after December 31<sup>st</sup> of the calendar year in which the Participant experienced a Separation from Service. Payments due on January 31<sup>st</sup> of a calendar year were determined on the previous December 31<sup>st</sup>, while payments due on the first business day of a calendar month were determined on the preceding January 31<sup>st</sup>).

(ii) Effective on and after October 2, 2017, a Participant who experienced or experiences a Separation from Service and elects to have his Post-2004 Subaccount distributed in Annual Payments shall have his Post-2004 Subaccount credited with earnings based on the rate of return in his selected Investment Options (or such Investment Options selected by the Investments Committee) until such time as the date upon which each Annual Payment is determined and paid.

(d) No provision of the Plan shall require the Company or the Employer to actually invest any amounts in any fund or in any other investment vehicle.

6.3 *Account Statements*. Each Participant will be given, at least annually, a statement showing (i) the amount of all Excess Salary Reduction Contributions, (ii) the amount of Employer Match, if any, made with respect to his Account for such calendar year, and (iii) the balance of the Participant's Account after crediting Investments.

## Section 7. Payment.

7.1 *Joint Payment Option Election*. (a) An Eligible Employee who has not elected or been deemed to have elected a Joint Payment Option under any other account balance plan that is required to be aggregated with the Plan under Code Section 409A shall, when he first becomes eligible to participate in the Plan, elect a Joint Payment Option on a written or electronic form acceptable to the Administrative Committee (a "Payment Election") at the same time that the Eligible Employee files his initial Deferral Election to commence participation in the Plan pursuant to Section 3.2, and in any event not later than his initial Election Filing Date. Such Payment Election shall be effective as of such initial Election Filing Date and shall be irrevocable. A Joint Payment Option elected pursuant to a Payment Election shall apply to all amounts credited to the Participant's Post-2004 Subaccount in the Plan and his Post-2004 Subaccount under any other account balance plan that is required to be aggregated with the Plan under Code Section 409A.

(b) (i) A Participant may elect to receive his entire Post-2004 Subaccount under either of the following Joint Payment Options: (A) a single lump sum; or, (B) annual payments over a period of two, three, four or five years ("Annual Payments"). The Annual Payments shall be treated as a single payment for purposes of this Section 7. If a Participant elects to receive Annual Payments over a period of two or more years, such Annual Payments shall be made in substantially equal annual payments, unless the Participant designates, at the time of making his Joint Payment Option election, a specific percentage of his Post-2004 Subaccount to be distributed in each year. All specified percentages must be a whole multiple of 10% and the total of all designated percentages must be equal to 100%. Effective as of October 2, 2017, if a Participant elects to receive Annual Payments over a period of two or more years, such Annual Payments shall be made in substantially equal annual and the Participant shall not be able to designate a specific percentage of his Post-2004 Subaccount to be distributed in each years.

(ii) If a Participant makes a Joint Payment Option Election to receive Annual Payments, the first payment shall be made on the later of (A) January 31 of the calendar year immediately following the calendar year in which the Participant experiences a Separation from Service or (B) the first business day of the seventh calendar month in which the Participant experiences a Separation from Service, and any subsequent Annual Payments shall be made on each applicable January 31<sup>st</sup> thereafter.

(iii) If a Participant makes a Joint Payment Option Election to receive payments in a single lump sum, such lump sum payment shall be made on the later of (A) January 31<sup>st</sup> of the calendar year immediately following the calendar year in which the Participant experiences a Separation from Service or (B) the first business day of the seventh calendar month following the calendar month in which the Participant experiences a Separation from Service. Alternatively, a Participant may elect for the single lump sum to be paid on January 31<sup>st</sup> of the second, third, fourth, or fifth calendar year following the end of the calendar year in which the Participant experiences a Separation from Service.

(iv) If a Participant does not make a Joint Payment Option Election in accordance with the terms of the Plan or under any other account balance plan that is required to be aggregated with the Plan under Code Section 409A, such Participant shall be deemed to have made a Joint Payment Option Election to receive his Post-2004 Subaccount in a single lump sum payable in accordance with the first sentence of Section 7.1(b)(iii).

(v) The following examples illustrate the provisions of this Section 7.1(b)

<u>Example 1</u>: Assume that a Participant elects (or is deemed to elect) a Joint Payment Option that provides for a single lump sum payment on the later of (A) January 31<sup>st</sup> of the calendar year following the calendar year in which he incurs a Separation from Service or (B) the first business day of the seventh calendar month following the calendar month in which the Participant experiences a Separation from Service, and the Participant experiences a Separation from Service on March 15, 2021. The lump sum shall be paid on January 31, 2022. The Participant alternatively could have elected to receive his lump sum payment on January 31, 2023, 2024, 2025 or 2026.

Example 2: Same facts as Example 1, except the Participant experiences a Separation from Service on September 15, 2021. In this example, the lump sum will be paid on the first business day in April 2022.

Example 3: If a Participant elects a Joint Payment Option that provides for Annual Payments over a period of four years in the event of a Separation from Service and experiences a Separation from Service on March 15, 2021, each payment on January 31, 2022 through 2025 will be comprised of approximately 25% of the Participant's Post-2004 Subaccount as of the Participant's date of Separation from Service, though the actual amount of each payment may not be the same due to crediting of investment gains and losses through December 31<sup>st</sup> of the calendar year prior to the calendar year of each such payment, or, effective after October 2, 2017, the actual amount of each payment may not be the same due to crediting of investment gains and losses through the payment date . Prior to October 2, 2017, a Participant alternatively could designate that 10% of his Post-2004 Subaccount be distributed on January 31, 2010, 20% on January 31, 2011, 30% on January 31, 2012 and 40% on January 31, 2013, or, any other combination of percentages that totals 100%.

Example 4: Same facts as Example 3, except the Participant experiences a Separation from Service on September 15, 2021. In this example, the first payment shall be made on the first business day in April 2022, and the remaining three payments will be made on January 31, 2023, 2024, and 2025. The alternative schedule described in Example 3 would result in payment of 10% of his Post-2004 Subaccount on the first business day in April 2022, 20% on January 31, 2023, 30% on January 31, 2024 and 40% on January 31, 2025.

7.2 *Payment on Account of Separation from Service*. If a Participant experiences a Separation from Service prior to his death, the Participant shall commence receiving payments from his Post-2004 Subaccount in accordance with the Joint Payment Option Election in effect with respect to the Participant.

7.3 *Payment on Account of Participant's Death.* If a Participant dies prior to his Separation from Service, or after his Separation from Service but prior to the distribution of his entire Post-2004 Subaccount, the Participant's entire Post-2004 Subaccount shall be paid to the Participant's beneficiary in a single lump sum payment within 90 days of the Participant's death. The Participant's Post-2004 Subaccount shall continue to be credited with earnings in accordance with Section 6.2 until his entire Post-2004 Subaccount is distributed.

#### Section 8. Unforeseeable Emergency Distributions and Deferral Revocations.

A Participant may request the Administrative Committee to accelerate distribution of all or any part of the value of his Post-2004 Subaccount solely for the purpose of alleviating an Unforeseeable Emergency. Payments of amounts as a result of an Unforeseeable Emergency may not exceed the amount necessary to satisfy such Unforeseeable Emergency, plus amounts necessary to pay taxes reasonably anticipated as a result of the distribution, and after taking into account any additional compensation that is available to the Participant upon cancellation of the Participant's Excess Salary Reduction Contributions. The Administrative Committee may request that the Participant provide certifications and other evidence of qualification for such Unforeseeable Emergency distribution as it determines appropriate. The decision of the Administrative Committee with respect to the grant or denial of all or any part of such request shall be in the sole discretion of the Administrative Committee, even if the Participant demonstrates that an Unforeseeable Emergency exists, and shall be final and binding and not subject to review. If a Participant receives a distribution upon an Unforeseeable Emergency pursuant to this Section 8 or a hardship withdrawal under the ViacomCBS 401(k) Plan (or its predecessor, the CBS 401(k) Plan), the Participant's Deferral Election will be canceled in its entirety for the remainder of the calendar year in which such Unforeseeable Emergency distribution is made under the Plan and under any other account balance plan that is required to be aggregated with the Plan under Code Section 409A.

#### Section 9. Beneficiary Designation.

A Participant's beneficiary designation for the Plan will automatically be the same as the Participant's beneficiary designation recognized under the ViacomCBS 401(k) Plan (or its predecessors, as applicable), unless a separate written designation of beneficiary form for the Plan has been properly filed with the Administrative Committee in a form acceptable to the Administrative Committee. In the absence of such a designation and at any other time when there is no existing beneficiary designated hereunder, the beneficiary of the Participant for payment of his Post-2004 Subaccount hereunder shall be the estate of the Participant. If two or more persons designated as a Participant's beneficiary are in existence with respect to his Post-2004 Subaccount, the amount of any lump sum payment payable hereunder shall be divided equally among such persons unless the Participant's beneficiary designation specifically provides for a different allocation.

#### Section 10. Nature of Interest of Participant.

Participation in the Plan will not create, in favor of any Participant, any right or lien in or against any of the assets of the Company or any Employer, and all amounts of compensation deferred hereunder shall at all times remain an unrestricted asset of the Company or the Employer. A Participant's rights to benefits payable under the Plan are not subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, or encumbrance. All payments hereunder shall be paid in cash from the general funds of the Company or applicable Employer and no special or separate fund shall be established and no other segregation of assets shall be made to assure the payment of benefits hereunder. Nothing contained in the Plan, and no action taken pursuant to its provisions, shall create or be construed to create a trust of any kind, or a fiduciary relationship, between any Employer and a Participant or any other person, and the Company's and each Employer's promise to pay benefits hereunder shall at all times remain unfunded as to the Participant.

#### Section 11. Administration.

11.1 *Administrative Committee*. The Plan shall be administered by the Administrative Committee. The Administrative Committee shall have sole and absolute discretion to interpret, where necessary, the provisions of the Plan (including, without limitation, by supplying omissions from, correcting deficiencies in, or resolving inconsistencies or ambiguities in, the language of the Plan), to determine the rights and status under the Plan of any Participant and other persons, to resolve questions or disputes arising under the Plan and to make any determinations with respect to the benefits hereunder and the persons entitled thereto as may be necessary for the purposes of the Plan.

11.2 *Powers of the Administrative Committee.* In furtherance of, but without limiting Section 11.1, the Administrative Committee shall have the following specific authorities, which it shall discharge in its sole and absolute discretion in accordance with the terms of the Plan (as interpreted, to the extent necessary, by the Administrative Committee):

(i) to determine who are Eligible Employees for purposes of participation in the Plan;

(ii) to interpret the terms and provisions of the Plan and to determine any and all questions arising under the Plan, including without limitation, the right to remedy possible ambiguities, inconsistencies, or omissions by a general rule or particular decision;

- (iii) to adopt rules consistent with the Plan;
- (iv) to approve certain amendments to the Plan;
- (v) to determine the amounts payable to any person under the Plan; and
- (vi) to conduct the claims procedure specified in Section 11.3.

## 11.3 Claims Procedure.

(a) <u>Initial Claim.</u> The Administrative Committee will make all determinations as to the right of any persons to benefits under the Plan in accordance with the governing Plan documents. Any denial by the Administrative Committee of a claim for benefits under the Plan by a Participant will be stated in writing by the Administrative Committee and delivered or mailed to the Participant within a reasonable period of time, but not later than 90 days after receipt of the claim by the Plan, unless the Administrative Committee determines that special circumstances require an extension of time for processing the claim. Written notice of the extension shall be furnished to the Participant prior to the termination of the initial 90-day period. The extension notice shall indicate the special circumstances requiring an extension of time and the date by which the Administrative Committee expects to render the benefit determination, which cannot exceed a period of 90 days from the end of the initial period.

(b) <u>Manner and Content of Notification of Benefit Determination</u>. The Administrative Committee shall provide a Participant with written notification (which may be delivered electronically) of any adverse benefit determination. The notification shall set forth in a manner calculated to be understood by the Participant:

(i) The specific reason or reasons for the adverse determination;

(ii) Reference to the specific Plan provisions on which the determination is based;

(iii) A description of any additional material or information necessary for the Participant to perfect the claim and an explanation of why such material or information is necessary; and

(iv) A description of the Plan's review procedures and the time limits applicable to such procedures, including a statement of the Participant's right to bring a civil action under Section 502(a) of ERISA following an adverse benefit determination on review.

(c) <u>Review of Benefit Determination</u>. The Administrative Committee will provide to any Participant whose claim for benefits has been denied an opportunity for a full and fair review of the denial. As part of the review, the Administrative Committee will:

(i) Provide a Participant at least 60 days (180 days for a claim regarding Disability) following the receipt of a notification of an adverse benefit determination within which to appeal the determination;

(ii) Provide a Participant the opportunity to submit written comments, documents, records, and other information relating to the claim for benefits;

(iii) Provide that a Participant shall be provided, upon request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to the Participant's claim for benefits; and

(iv) Provide for a review that takes into account all comments, documents, records, and other information submitted by the Participant relating to the claim, without regard to whether such information was submitted or considered in the initial benefit determination.

(d) <u>Notification of Determination on Review</u>. The Administrative Committee shall provide a Participant with written notification (which may be delivered electronically) of the Plan's benefits determination on review within a reasonable period of time, but not later than 60 days after receipt of the claim by the Plan, unless the Administrative Committee determines that special circumstances require an extension of time for processing the claim. Written notice of the extension will be furnished to a Participant prior to the termination of the initial 60-day period. The extension notice will indicate the special circumstances requiring an extension of time and the date by which the Plan expects to render the benefit determination on review, which cannot exceed a period of 60 days from the end of the initial period. In the case of an adverse benefit determination, the notification shall set forth, in a manner calculated to be understood by the Participant:

(i) The specific reason or reasons for the adverse determination;

(ii) Reference to the specific Plan provisions on which the benefit determination is based;

(iii) A statement that the Participant is entitled to receive, upon request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to the Participant's claim for benefits; and

(iv) A statement describing any voluntary appeal procedures offered by the Plan and the Participant's right to obtain the information about such procedures and a statement of the Participant's right to bring an action under Section 502(a) of ERISA.

11.4 *Finality of Administrative Committee Determinations and Delegation*. Determinations by the Administrative Committee and any interpretation, rule, or decision adopted by the Administrative Committee under the Plan or in carrying out or administering the Plan shall be final and binding for all purposes and upon all interested persons, their heirs, and personal representatives. The Administrative Committee shall be the named fiduciary of the Plan. The Administrative Committee may delegate to any person any one or more of its powers, functions, duties or responsibilities with respect to the Plan, including, without limitation, duties with respect to the processing, review, investigation, approval and payment of Accounts.

11.5 Rules *and Regulations Established by Administrative Committee*. The Administrative Committee may promulgate any rules and regulations it deems necessary in order to carry out the purposes of the Plan or to interpret the terms and conditions of the Plan; provided however, that no rule, regulation or interpretation shall be contrary to the provisions of the Plan. The rules, regulations and interpretations made by the Administrative Committee shall, subject only to the claims procedure outlined in Section 11.3, be final and binding on any employee, former employee, or other individual making a claim for Plan benefits.

#### Section 12. No Employment Rights.

No provisions of the Plan or any action taken by the Company, any Employer, the Board of Directors, or the Administrative Committee shall give any person any right to be retained in the employ of the Company or any Employer, and the right and power of the Company or any Employer to dismiss or discharge any Participant is specifically reserved.

#### Section 13. Amendment, Suspension, and Termination.

The Administrative Committee shall have the right to amend the Plan at any time, unless provided otherwise in the Company's governing documents. The Board of Directors shall have the right to suspend or terminate the Plan at any time. No amendment, suspension or termination shall, without the consent of a Participant, adversely affect such Participant's rights in his Account; provided, however, that the consent requirement of Participants to certain actions shall not apply to any amendment or termination that is deemed necessary by the Company to avoid the imposition on any person of additional taxes, penalties or interest under Code Section 409A. In the event the Plan is terminated, the Administrative Committee may continue to administer the Plan in accordance with the relevant provisions thereof or shall have the right to change the time and form of distribution of Participants' Accounts, including requiring that the Accounts be immediately distributed in the form of a lump sum payment, provided, however, that no such change in the time or form of payment shall cause the Plan to fail to comply with the requirements of Code section 409A.

#### Section 14. Miscellaneous.

14.1 *Severability*. If a provision of the Plan shall be held invalid, the invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the invalid provision had not been included in the Plan.

14.2 *Governing Law.* The provisions of the Plan shall be governed by and construed in accordance with the laws of the State of New York, to the extent not preempted by the laws of the United States.

14.3 *Gender*. Wherein used herein, words in the masculine form shall be deemed to refer to females as well as males.

14.4 *Code Section 409A.* To the extent applicable, it is intended that the Plan comply with the provisions of Code Section 409A. References to Code Section 409A shall include any proposed, temporary or final regulation, or any other guidance, promulgated with respect to such section by the U.S. Department of the Treasury or the Internal Revenue Service. The Plan shall be administered and interpreted in a manner consistent with this intent. If any provision of the Plan is susceptible of two interpretations, one of which results in the compliance of the Plan with Code Section 409A and the applicable Treasury Regulations, and one of which does not, then the provision shall be given the interpretation that results in compliance with Code Section 409A and the applicable Treasury Regulations. Notwithstanding the foregoing or any other provision of the Plan to the contrary, neither the Company nor any of its subsidiaries or affiliates shall be deemed to guarantee any particular tax result for any Participant, spouse, or beneficiary with respect to any payments provided hereunder.

IN WITNESS WHEREOF, in accordance with the Administrative Committee's August 20, 2021 Unanimous Written Consent, the Company has caused this Plan to be executed by its duly authorized representative this 20th day of September, 2021.

ViacomCBS Inc.

By:	/s/ Mark Beatty	
	Mark Beatty	

\_\_\_\_\_

Its: Member, ViacomCBS Administrative Committee

#### AMENDMENT NO. 1 TO THE AMENDED AND RESTATED VIACOMCBS EXCESS 401(k) PLAN-PART A (2021 RESTATEMENT) & PART B (2021 RESTATEMENT)

Effective as of February 16, 2022, the ViacomCBS Excess 401(k) Plan (each of Part A and Part B) is amended such that, for all periods on or after February 16, 2022, all references to "ViacomCBS" or "ViacomCBS Inc." shall be deemed to refer to "Paramount Global", including such that: all references to the "ViacomCBS Administrative Committee" or "ViacomCBS Investments Committee" shall be deemed to refer to "Paramount Global Administrative Committee" and "Paramount Global Investments Committee", respectively; all references to "ViacomCBS 401(k) Plan" shall be deemed to refer to "Paramount Global 401(k) Plan"; and reference to "ViacomCBS Inc." in the definition of "Company" shall be deemed to refer to "Paramount Global".

IN WITNESS WHEREOF, pursuant to a resolution of the Paramount Global Administrative Committee, the undersigned hereby executes this amendment this 7<sup>th</sup> day of October, 2022.

By: /s/ Mark Beatty

Name: Mark Beatty

Its: Member, Paramount Global Administrative Committee

#### AMENDMENT NO. 1 TO THE AMENDED AND RESTATED VIACOMCBS EXCESS 401(k) PLAN FOR DESIGNATED SENIOR EXECUTIVES-PART A (2021 RESTATEMENT) & PART B (2021 RESTATEMENT)

Effective as of February 16, 2022, the ViacomCBS Excess 401(k) Plan for Designated Senior Executives (each of Part A and Part B) is amended such that, for all periods on or after February 16, 2022, all references to "ViacomCBS" or "ViacomCBS Inc." shall be deemed to refer to "Paramount Global", including such that: all references to the "ViacomCBS Administrative Committee" or "ViacomCBS Investments Committee" shall be deemed to refer to "Paramount Global Investments Committee", respectively; all references to "ViacomCBS 401(k) Plan" shall be deemed to refer to "Paramount Global 401(k) Plan"; and reference to "ViacomCBS Inc." in the definition of "Company" shall be deemed to refer to "Paramount Global".

IN WITNESS WHEREOF, pursuant to a resolution of the Paramount Global Administrative Committee, the undersigned hereby executes this amendment this 7<sup>th</sup> day of October, 2022.

By: /s/ Mark Beatty

Name: Mark Beatty

Its: Member, Paramount Global Administrative Committee (2021 Restatement)

# Table of Contents

Section	1. Establishment and Purpose of the Plan	3		
1.1	<u>Establishment</u>	3		
1.2	Purpose	3		
Section 2	2. Definitions	3		
Section 3	3. <u>Participation</u>	5		
3.1	Election to Participate	5		
3.2	Amount of Elections	5		
Section 4. Individual Account				
4.1	Creation of Accounts	5		
4.2	Joint Payment Option Election	5		
4.3	Investments	7		
4.4	Account Statements	8		
Section 5. <u>Payment</u>				
5.1	Payment on Account of Termination of Employment for Reasons Other than Disability	8		
5.2	Payment on Account of Disability	8		
Section (	6. Nature of Interest of Participant	9		
Section '	7. Hardship Distributions and Deferral Revocations	9		
Section 8	8. <u>Beneficiary Designation</u>	9		
Section 9	9. <u>Administration</u>	10		
9.1	Administrative Committee	10		
9.2	Powers of the Administrative Committee	10		
9.3	Claims Procedure	10		
9.4	Finality of Administrative Committee Determinations	10		
9.5	<u>Severability</u>	10		
9.6	Governing Law	10		
9.7	Gender	11		
Section	10. <u>No Employment Rights</u>	11		
Section	11. Amendment, Suspension, and Termination	11		

### Section 1. Establishment and Purpose of the Plan.

1.1 *Establishment.* The Viacom Bonus Deferral Plan was adopted as of August 28, 2002 as an unfunded plan of voluntarily deferred compensation for the benefit of Participants. As of December 31, 2005, it was renamed the CBS Bonus Deferral Plan, and as of October 1, 2021, it is hereby amended and restated as the ViacomCBS Bonus Deferral Plan. Any Eligible Employee who is identified by the Company on or after August 28, 2002 as a reporting person for purposes of Section 16 of the Securities Exchange Act of 1934 ("Reporting Employees") or any employee of an Employer who is eligible to participate in the CBS Bonus Deferral Plan, or as applicable its successors, including the ViacomCBS Bonus Deferral Plan, and whose securities may be attributable to a Reporting Employee for purposes of Section 16 of the Securities Exchange Act of 1934 shall no longer be eligible to participate in this Plan, and shall instead be eligible to participate in the CBS Bonus Deferral Plan for Designated Senior Executives, or as applicable its successors, including the ViacomCBS Bonus Deferral Plan for Designated Senior Executives (the "Executive Bonus Deferral Plan"). Any deferrals made under the Plan by any Reporting Employee who was a participant in the Plan on August 28, 2002 and by any Reporting Employee (or any other Eligible Employee whose securities may be attributable to a Reporting Employee) prior to the date he becomes a Reporting Employee (or the date his securities are attributable to a Reporting Employee) were transferred to the Executive Bonus Deferral Plan as of December 1, 2005 or, if later, as of the date he becomes a Reporting Employee).

1.2 *Purpose.* The purpose of Part A of this Plan is to provide the means by which an Eligible Employee could have, in certain circumstances, elected to defer receipt of a portion of his cash bonus paid under the CBS Corporation Short Term Incentive Plan and any other comparable annual cash bonus plan sponsored by any Employer prior to January 1, 2005. All deferrals on or after January 1, 2005, are governed by Part B of the Plan.

### Section 2. Definitions.

The following words and phrases as used in Part A of this Plan have the following meanings:

2.1 The term "Account" shall mean a Participant's individual account, as described in Section 4 of the Plan.

2.2 The term "Administrative Committee" means (i) for the periods prior to June 1, 2020, the CBS Retirement Committee, and (ii) for the periods on and after June 1, 2020, the ViacomCBS Administrative Committee. The Administrative Committee may act on its own behalf or through the actions of its duly authorized delegate or delegates.

2.3 The term "Board of Directors" means the Board of Directors of the Company.

2.4 The term "Bonus" means any cash bonus paid under the CBS Corporation Short-Term Incentive Plan and any other comparable annual cash bonus plan sponsored by any Employer.

2.5 The term "Bonus Deferral Contributions" means the portion of the Participant's Bonus that he elected to defer under the terms of Part A of this Plan. The portion of any Bonus earned in the year 2002 that an Eligible Employee elected to defer under the CBS Excess 401(k) Plan shall be deferred under this Plan, and shall not be recognized under the CBS Excess 401(k) Plan.

2.6 The term "Company" shall refer to (i) CBS Corporation and its subsidiaries for periods prior to December 4, 2019, and (ii) ViacomCBS Inc. for the periods on and after December 4, 2019.

2.7 The term "Disability" or "Disabled" means that a Participant (i) has been determined to be disabled by the Social Security Administration, or (ii) is receiving benefits under the provisions of the long-term disability plan covering such Participant that is sponsored by or participated in by the Participant's Employer. The date a Participant meets the definition of Disability shall be treated as the date he terminates employment for purposes of Section 5 of the Plan.

2.8 The term "Eligible Employee" means an employee of an Employer who is an eligible employee under the CBS Excess 401(k) Plan, and as applicable its successors. If an employee becomes an Eligible Employee in any Plan Year, such employee shall remain an Eligible Employee for all future Plan Years during which the Eligible Employee remains an eligible employee under the CBS 401(k) Excess Plan, and as applicable its successors. In no event shall any Reporting Employee be considered an Eligible Employee under this Plan.

2.9 The term "Employer" means the Company and any affiliate or subsidiary that adopts the Plan on behalf of its Eligible Employees.

2.10 The term "Investments Committee" means (i) for the periods prior to June 1, 2020, the Investments Committee for CBS Defined Contribution Plans, and (ii) for the periods on and after June 1, 2020, the ViacomCBS Investments Committee.

2.11 The term "Investment Options" means (i) prior to January 1, 2014, the investment funds available to participants in the CBS 401(k) Plan, excluding the Self-Directed Brokerage Account, (ii) effective as of January 1, 2014 through December 31, 2014, the notional investment options selected by the CBS Retirement Committee in its sole discretion, and (iii) effective as of January 1, 2015, means the notional investment options elected by the Investments Committee, in its sole discretion.

2.12 The term "Joint Payment Option" means the Participant's joint payment option election in accordance with Section 4.2 with respect to the distribution upon his termination of employment of amounts credited to his account in the CBS Excess 401(k) Plan, or as applicable its successors, including the ViacomCBS Excess 401(k) Plan, to his Account in this Plan.

2.13 The term "Participant" means an Eligible Employee who elects to have Bonus Deferral Contributions made to the Plan.

2.14 The term "Plan" means (i) for the periods prior to October 1, 2021, the CBS Bonus Deferral Plan, and (ii) for the periods on and after October 1, 2021, the ViacomCBS Bonus Deferral Plan.

### Section 3. Participation.

- 3.1 *Election to Participate.* 
  - (a) Prior to January 1, 2005, an Eligible Employee must have elected to participate in the Plan.

(b) Any election to defer a portion of a Bonus earned in the year 2002 that was made by an Eligible Employee prior to August 28, 2002 under the CBS Excess 401(k) Plan shall be recognized by and be deemed to have been made under this Plan, and such Eligible Employee shall have become a Participant in this Plan on August 28, 2002.

(c) For the Plan Year in which an employee first became an Eligible Employee, such Eligible Employee must have elected to make a Bonus Deferral Contribution with respect to any Bonus scheduled to be paid in the next succeeding calendar year within 30 days of the date he first became an Eligible Employee in order for the election to be valid. Prior to December 31 of each Plan Year, an Eligible Employee could have elected to make a Bonus Deferral Contribution with respect to any Bonus scheduled to be paid in the second succeeding calendar year. For example, prior to December 31, 2002, an Eligible Employee may have made a Bonus Deferral Contribution election with respect to any cash bonus to be earned in 2003 that was scheduled to be paid in 2004 under the CBS Corporation Short-Term Incentive Plan. An Eligible Employee could have made an Excess Bonus Deferral Contribution election. On and after January 1, 2005, all Bonus Deferral Contributions are deferred under Part B of the Plan.

3.2 *Amount of Elections.* Each election filed by a Participant must have specified the amount of Bonus Deferral Contribution in a whole percentage between 1% and 15% of the Participant's applicable Bonus.

## Section 4. Individual Account.

4.1 *Creation of Accounts.* The Company will maintain an Account in the name of each Participant. Each Participant's Account will be credited with the amount of the Participant's Bonus Deferral Contributions made in all Plan Years, including any Bonus Deferral Contributions for the Bonus earned in the year 2002 that are attributable to Bonus Deferral Contribution elections originally made under the CBS Excess 401(k) Plan.

### 4.2 *Joint Payment Option Election.*

(a) With respect to each Participant in the Plan on August 28, 2002 who was immediately prior to that date a participant in the CBS Excess 401(k) Plan, any Joint Payment Option election recognized under the CBS Excess 401(k) Plan shall apply to the total amount credited to the Participant's Account in this Plan, together with the total amounts credited to the Participant's account in the CBS Excess 401(k) Plan.

(b) (i) If an Eligible Employee who first became a Participant in this Plan after August 28, 2002 and was a Participant in the CBS Excess 401(k) Plan, any Joint Payment Option election recognized at such time under the CBS Excess 401(k) Plan regarding any Excess Salary Deferral Contributions made by the Participant under the CBS Excess 401(k) Plan shall apply to the Participant's Account in this Plan, together with the total amounts credited to the Participant's account in the CBS Excess 401(k) Plan.

(ii) If such Eligible Employee was not a participant in the CBS Excess 401(k) Plan and, therefore, did not have in effect a Joint Payment Option election under such Plan, the Eligible Employee shall elect a Joint Payment Option under this Plan at the same time that the Eligible Employee files his initial election to commence participation in the Plan pursuant to Section 3.2. Any such Joint Payment Option election made by the Participant in this Plan shall also apply to any future Excess Salary Deferral Contributions that the Participant may make under the CBS Excess 401(k) Plan, or its successors, including the ViacomCBS Excess 401(k) Plan.

(c) A Participant may elect to receive his entire Account in either of the following Joint Payment Options: (l) a single lump sum; or, (2) annual payments over a period of two, three, four or five years in annual payments on or about January 31<sup>st</sup> beginning in the calendar year immediately following the end of the Plan Year in which the Participant terminates employment. If no Joint Payment Option election is made in accordance with the terms of the Plan, a Participant shall be deemed to have elected to receive his Account in a single lump sum on or about January 31<sup>st</sup> of the calendar year immediately following the end of the Plan Year in which the Participant terminates employment. If a Participant makes a Joint Payment Option election to receive payments in a single lump sum, such lump sum shall be payable on or about January 31<sup>st</sup> of the calendar year immediately following the end of the Plan Year in which the Participant terminates employment, unless the Participant elects to be paid on or about January 31<sup>st</sup> of the second, third, fourth, or fifth calendar year following the year in which the Participant terminates employment. In the event a Participant elects to receive annual payments over a period of two or more years, such annual payments shall be made in substantially equal annual payments, unless the Participant designates at the time of making his Joint Payment Option election a specific percentage of his Account to be distributed in each year. All specified percentages must be a whole multiple of 10% and the total of all designated percentages must be equal to 100%.

Example 1: If a Participant elects (or is deemed to elect) a Joint Payment Option that provides for a lump sum payment and terminates employment in 2022, such lump sum shall be paid on or about January 31, 2023. A Participant alternatively could designate January 31<sup>st</sup> of 2024, 2025, 2026 or 2027 in which to receive his lump sum.

Example 2: If a Participant elects a Joint Payment Option that provides for annual payments over a period of four years and terminates employment in 2022, each payment paid on or about January 31<sup>st</sup>, 2023 through 2026 will be comprised of approximately 25% of the Participant's Account as of the Participant's date of termination. A Participant alternatively could designate 10% of his Account to be distributed in January 2023, 20% in January 2024, 30% in January 2025 and 40% in January 2026; or, any other combination of percentages which totals 100%.

(e) Any change of Joint Payment Option election made by a Participant under the CBS Excess 401(k) Plan, or its successors including the ViacomCBS Excess 401(k) Plan, shall apply to the Participant's Account in this Plan. A Participant may change his Joint Payment Option no more than three times over the course of his employment with the Company or any affiliate. A Participant may change an existing Joint Payment Option election only one time in any calendar year. Any change of a Participant's existing Joint Payment Option election made less than six months prior to the Participant's termination of employment for any reason shall be null and void and the Participant's last valid Joint Payment Option shall remain in effect.

#### 4.3 Investments.

(a) Prior to January 1, 2014, all Bonus Deferral Contributions were credited through December 31<sup>st</sup> of the calendar year in which the Participant terminates employment with an amount equal to such amount which would have been earned had such contributions been invested in the same Investment Options and in the same proportion as the Participant may elect, from time to time, to have his Salary Reduction Contributions and Matching Employer Contributions invested under the CBS 401(k) Plan, or its successors including the ViacomCBS 401(k) Plan; or if no such election has been made, in the PRIMCO Stable Value Fund (or any successor fund).

Effective as of January 1, 2014, a Participant may select from a list of notional Investment Options how the balance of his Account will be invested. If no selection is made, the Participant's Account will be notionally invested in the "qualified default investment alternative" (within the meaning of CBS 401(k) Plan, or its successors including the ViacomCBS 401(k) Plan) in effect from time to time. Earnings and losses received on the Participant's notional investments will be credited to the Participant's Account in the manner designated by the Investments Committee. The Investments Committee shall develop such procedures as it, in its discretion, deems advisable with respect to the selection of notional investments by Participants and the reflection of value attributable to such notional investments in their Accounts, including, without limitation, procedures which restrict a Participant's ability to notionally invest in certain Investment Options.

(b) (i) Prior to October 2, 2017, if a terminated Participant elected (or was deemed to elect) a single lump sum Joint Payment Option payable in the first calendar year following the calendar year in which the Participant terminated employment, no additional adjustments were made to the Participant's Account after December 31st of the calendar year in which the Participant terminated employment. If a Participant elected a single lump sum Joint Payment Option payable in the second, third, fourth or fifth calendar year following the calendar year in which the Participant terminated employment, the Participant's Account was credited with earnings based on the rate of return in the PRIMCO Stable Value Fund (or any successor fund) beginning January 1st of the calendar year following the year in which the Participant terminated employment and continuing through December 31<sup>st</sup> of the calendar year immediately preceding the calendar year in which the single lump sum was paid.

(ii) Effective October 2, 2017, if a terminated Participant elected (or is deemed to elect) a single lump sum Joint Payment Option payable in the first calendar year following the calendar year in which the Participant terminates employment, or if the Participant elects a single lump sum Joint Payment Option payable in the second, third, fourth or fifth calendar year following the calendar year in which the Participant terminates employment, the Participant's Account shall be credited with earnings based on the rate of return of his selected Investment Options (or such Investment Options selected by the Investments Committee) until such time as the date upon which the single lump sum payment is determined and paid.

(c) (i) Prior to October 2, 2017, if a terminated Participant elected annual payments, no additional adjustments were made to any amount payable in the first calendar year following the year in which the Participant terminated employment. For any annual payments made in the second, third, fourth or fifth year following the calendar year in which the Participant terminated employment, the Participant's Account was credited with earnings based on the rate of return in the PRIMCO Stable Value Fund (or any successor fund) beginning January 1st of the calendar year following the year in which the Participant terminated employment and continuing through December 31st of the calendar year immediately preceding the calendar year in which each payment was made.

(ii) Effective October 2, 2017, if a terminated Participant elects annual payments, the Participant's Account shall be credited with earnings based on the rate of return in his selected Investment Options (or such Investment Options selected by the Investments Committee) until such time as the date upon which each annual payment is determined and paid.

(d) No provision of this Plan shall require the Company or the Employer to actually invest any amounts in any fund or in any other investment vehicle.

4.4 *Account Statements*. Each Participant will be given, at least annually, a statement showing (i) Bonus Deferral Contributions, and (ii) the balance of the Participant's Account after crediting Investments.

#### Section 5. Payment.

5.1 *Payment on Account of Termination of Employment for Reasons Other than Disability.* A Participant (or a Participant's beneficiary) shall be paid the balance in his Account following termination of employment in accordance with the Joint Payment Option in effect with respect to the Participant.

5.2 Payment on Account of Disability. A Participant (or a Participant's beneficiary) shall be paid the balance in his Account following the date he meets the definition of Disability in accordance with the Joint Payment Option in effect with respect to the Participant. If a Participant no longer meets the definition of Disability and returns to work with an Employer, no further payments shall be made on account of the prior Disability, and distribution of his remaining Account shall be made as otherwise provided in this Section 5 at the time of his subsequent termination of employment.

#### Section 6. Nature of Interest of Participant.

Participation in this Plan will not create, in favor of any Participant, any right or lien in or against any of the assets of the Company or any Employer, and all amounts of Compensation deferred hereunder shall at all times remain an unrestricted asset of the Company or the Employer. A Participant's rights to benefits payable under the Plan are not subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, or encumbrance. All payments hereunder shall be paid in cash from the general funds of the Company or applicable Employer and no special or separate fund shall be established and no other segregation of assets shall be made to assure the payment of benefits hereunder. Nothing contained in this Plan, and no action taken pursuant to its provisions, shall create or be construed to create a trust of any kind, or a fiduciary relationship, between any Employer and a Participant or any other person, and the Company's and each Employer's promise to pay benefits hereunder shall at all times remain unfunded as to the Participant.

### Section 7. Hardship Distributions and Deferral Revocations.

A Participant may request the Administrative Committee to accelerate distribution of all or any part of the value of his Account solely for the purpose of alleviating an immediate financial emergency. For purposes of the Plan, such an immediate financial emergency shall mean an unanticipated emergency that is caused by an event beyond the control of the Participant and which would result in severe financial hardship to the Participant if early distribution were not permitted. The Administrative Committee may request that the Participant provide certifications and other evidence of qualification for such emergency hardship distribution as it determines appropriate. The decision of the Administrative Committee with respect to the grant or denial of all or any part of such request shall be in the sole discretion of the Administrative Committee, whether or not the Participant demonstrates an immediate financial emergency exists, and shall be final and binding and not subject to review.

### Section 8. Beneficiary Designation.

Prior to January 1, 2014, Participant's beneficiary designation for this Plan were automatically the same as the Participant's beneficiary designation recognized under the CBS Excess 401(k) Plan. Effective as of January 1, 2014, a Participant's beneficiary designation for the Plan will be a separate written designation in a form acceptable to the Administrative Committee that has been properly filed with the Administrative Committee and recorded in the Company's records. If no such beneficiary designation has been made under the Plan, then a Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation recognized under the CBS Excess 401(k) Plan, or its successors, including the ViacomCBS Excess 401(k) Plan, as applicable. If no such beneficiary designation has been made under the CBS 401(k) Plan, or its successors, then a Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan, or its successors, then a Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's beneficiary designation for the Plan shall be the same as the Participant's ben

#### Section 9. Administration.

9.1 *Administrative Committee*. This Plan will be administered by the Administrative Committee, the members of which will be selected by the Board of Directors.

9.2 *Powers of the Administrative Committee*. The Administrative Committee's powers will include, but will not be limited to, the power:

(i) to determine who are Eligible Employees for purposes of participation in the Plan;

(ii) to interpret the terms and provisions of the Plan and to determine any and all questions arising under the Plan, including without limitation, the right to remedy possible ambiguities, inconsistencies, or omissions by a general rule or particular decision;

- (iii) to adopt rules consistent with the Plan; and
- (iv) to approve certain amendments to the Plan.

9.3 *Claims Procedure.* The Administrative Committee shall have the exclusive right to interpret the Plan and to decide any and all matters arising thereunder. In the event of a claim by a Participant as to the amount of any distribution or method of payment under the Plan, within 90 days of the filing of such claim, unless special circumstances require an extension of such period, such person will be given notice in writing of any denial, which notice will set forth the reason for the denial, the Plan provisions on which the denial is based, an explanation of what other material or information, if any, is needed to perfect the claim, and an explanation of the claims review procedure. The Participant may request a review of such denial within 60 days of the date of receipt of such denial by filing notice in writing with the Administrative Committee. The Participant will have the right to review pertinent Plan documents and to submit issues and comments in writing. The Administrative Committee will respond in writing to a request for review within 60 days of receiving it, unless special circumstances require an extension of such period. The Administrative Committee, at its discretion, may request a meeting to clarify any matters deemed appropriate.

9.4 *Finality of Administrative Committee Determinations*. Determinations by the Administrative Committee and any interpretation, rule, or decision adopted by the Administrative Committee under the Plan or in carrying out or administering the Plan shall be final and binding for all purposes and upon all interested persons, their heirs, and personal representatives.

9.5 *Severability.* If a provision of the Plan shall be held illegal or invalid, the illegality or invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included in the Plan.

9.6 *Governing Law.* The provisions of the Plan shall be governed by and construed in accordance with the laws of the State of New York, to the extent not preempted by the laws of the United States.

9.7 *Gender.* Wherein used herein, words in the masculine form shall be deemed to refer to females as well as males.

## Section 10. No Employment Rights.

No provisions of the Plan or any action taken by the Company, the Board of Directors, or the Administrative Committee shall give any person any right to be retained in the employ of any Employer, and the right and power of the Company to dismiss or discharge any Participant is specifically reserved.

#### Section 11. Amendment, Suspension, and Termination.

The Administrative Committee shall have the right to amend the Plan at any time, unless provided otherwise in the Company's governing documents. The Board of Directors shall have the right to suspend or terminate the Plan at any time. No amendment, suspension or termination shall, without the consent of a Participant, adversely affect such Participant's rights in his account. In the event the Plan is terminated, the Administrative Committee shall continue to administer the Plan in accordance with the relevant provisions thereof.

IN WITNESS WHEREOF, in accordance with the Administrative Committee's August 20, 2021 Unanimous Written Consent, the Company has caused this Plan to be executed by its duly authorized representative this 20th day of September, 2021.

ViacomCBS Inc.

- By: /s/ Mark Beatty Mark Beatty
- Its: Member, ViacomCBS Administrative Committee

(2021 Restatement)

Section	1. Establishment and Purpose of the Plan	1
1.1	<u>Establishment</u>	1
1.2.	Amendment and Restatement	1
1.3	Merger	1
1.4	Reporting Employees	1
1.5	Purpose	2
Section	a 2. <u>Definitions</u>	2
Section	a 3. <u>Participation</u>	5
3.1	Election to Participate	5
3.2	Amount of Elections	6
Section	4. <u>Individual Account</u>	6
4.1	Creation of Accounts	6
4.2	Investments	6
4.3	Account Statements	8
Section	n 5. <u>Payment</u>	8
5.1	Joint Payment Option Election	8
5.2	Payment on Account of Separation from Service	10
5.3	Payment on Account of Participant's Death	10
Section	6. Unforeseeable Emergency Distributions and Deferral Revocations	10
Section	7. Beneficiary Designation	11
Section	8. Nature of Interest of Participant	11
Section	9. <u>Administration</u>	11
9.1	Administrative Committee	11
9.2	Powers of the Administrative Committee	11
9.3	Claims Procedure	12
9.4	Finality of Administrative Committee Determinations	12
Section	n 10. <u>No Employment Rights</u>	12
Section	11. Amendment, Suspension, and Termination	13
Section	n 12. <u>Miscellaneous</u>	13
12.1	<u>Severability</u>	13
12.2	Governing Law	13
12.3	Gender	13
12.4	Code Section 409A	13

### Section 1. Establishment and Purpose of the Plan.

1.1 *Establishment*. The Viacom Bonus Deferral Plan was adopted as of August 28, 2002 as an unfunded plan of voluntarily deferred compensation for the benefit of Participants. On December 31, 2005, the Viacom Bonus Deferral Plan was renamed the CBS Bonus Deferral Plan, and as of October 1, 2021, it was amended and restated and renamed the ViacomCBS Bonus Deferral Plan.

1.2. Amendment and Restatement. The Plan was restated, effective as of January 1, 2009, by the adoption of Part B of the Plan, as set forth herein, and is hereby amended and restated effective as of October 1, 2021. Part A of the Plan, consisting of the original Plan adopted August 28, 2002, along with certain amendments, applies to compensation that was Deferred during calendar years ending prior to January 1, 2005 in accordance with the terms of those documents in effect from time to time prior to October 3, 2004, subject to certain exceptions set forth in Part A of the Plan. The provisions of this Part B shall apply to compensation that is Deferred on or after January 1, 2005. This Part B of the Plan is intended to meet all of the requirements of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code"), so that Participants will be eligible to defer the receipt of, and the liability for the federal income tax with respect to, certain items of compensation from one year to a later year in accordance with the provisions of the Plan.

1.3 *Merger*. Under an Agreement and Plan of Merger between CBS Corporation and Viacom Inc., dated August 13, 2019, Viacom Inc. merged with and into CBS Corporation on December 4, 2019, to become ViacomCBS Inc.

1.4 Reporting Employees. Any Eligible Employee who is identified by the Company on or after August 28, 2002 as a reporting person for purposes of Section 16 of the Securities Exchange Act of 1934 ("Reporting Employee") or any employee of an Employer who is eligible to participate in the Plan and whose securities may be attributable to a Reporting Employee for purposes of Section 16 of the Securities Exchange Act of 1934 shall no longer be eligible to participate in the Plan, and shall instead be eligible to participate in the CBS Bonus Deferral Plan for Designated Senior Executives, or as of October 1, 2021, its successor the ViacomCBS Bonus Deferral Plan for Designated Senior Executives (the "Executive Bonus Deferral Plan") effective as of the date on which the Employee becomes a Reporting Employee (or the date his securities are attributable to a Reporting Employee). Any deferrals made under the Plan by any Reporting Employee who was a participant in the Plan on August 28, 2002 and by any Reporting Employee (or any other Eligible Employee whose securities may be attributable to a Reporting Employee) prior to the date he becomes a Reporting Employee (or the date his securities are attributable to a Reporting Employee) shall be transferred to the Executive Bonus Deferral Plan as of December 1, 2005 or, if later, as of the date the Employee becomes a Reporting Employee (or the date his securities are attributable to a Reporting Employee). Elections as to the time and form of payment made by a Reporting Employee (or an employee whose securities may be attributable to a Reporting Employee) under the Plan prior to the date his account is transferred to the Executive Bonus Deferral Plan shall remain in full force following the transfer.

1.5 *Purpose*. The purpose of Part B of the Plan is to provide a means by which a select group of Eligible Employees may, in certain circumstances, elect to defer receipt of a portion of their cash Bonuses received on or after January 1, 2005.

### Section 2. Definitions.

The following words and phrases as used in Part B of the Plan have the following meanings:

2.1 The term "Account" shall mean a Participant's individual account, as described in Section 4.1 of the Plan.

2.2 The term "Administrative Committee" means (i) for the periods prior to June 1, 2020, the CBS Retirement Committee, and (ii) for the periods on and after June 1, 2020, the ViacomCBS Administrative Committee. The Administrative Committee may act on its own behalf or through the actions of its duly authorized delegates.

2.3 The term "Annual Payments" is defined in Section 5.1 (b)(i).

2.4 The term "Board of Directors" means the Board of Directors of the Company.

2.5 The term "Bonus" means any cash bonus paid under the STIP and any other annual cash bonus plan (or annual component of a cash bonus plan) sponsored by an Employer which, in the discretion of the Administrative Committee, is comparable to the STIP.

2.6 The term "Bonus Deferral Contributions" means the portion of the Participant's Bonus that he elects to defer under the terms of the Plan. The portion of any Bonus earned in the year 2002 that an Eligible Employee elected to defer under the CBS Excess 401(k) Plan (formerly known as the Viacom Excess 401(k) Plan) shall be deferred under the Plan, and shall not be recognized under the CBS Excess 401(k) Plan, or its successor the ViacomCBS Excess 401(k) Plan.

2.7 The term "Code" means the Internal Revenue Code of 1986, as amended.

2.8 The term "Company" shall refer to (i) CBS Corporation and its subsidiaries for periods prior to December 4, 2019, and (ii) ViacomCBS Inc. for the periods on and after December 4, 2019.

2.9 The term "Deferral Election" is defined in Sections 3.1(b) and 3.1 (c).

2.10 The term "Deferred" means that an amount is considered to be deferred within the meaning of Treasury Regulations sections 1.409A-6(a)(2) and 1.409A-6(a)(3).

2.11 The term "Disability" or "Disabled" means that a Participant (i) has been determined to be disabled by the Social Security Administration or (ii) is receiving benefits under the provisions of the long-term disability plan covering such Participant that is sponsored by or participated in by the Participant's Employer.

2.12 The term "Election Agreement" is defined in Section 3.1(d).

2.13 The term "Election Filing Date" means, except as provided in Section 3.1(c), the date not later than the December 31 immediately preceding the first day of the applicable calendar year for which a particular Deferral Election is made.

2.14 The term "Eligible Employee" means an employee of an Employer who is an eligible employee under (i) for the periods prior to October 1, 2021, the CBS Excess 401(k) Plan, and (ii) for the periods on or after October 1, 2021, the ViacomCBS Excess 401(k) Plan. If an employee becomes an Eligible Employee in any calendar year, such employee shall remain an Eligible Employee for all future calendar years during which the Eligible Employee remains an eligible employee under the CBS Excess 401(k) Plan, or its successor the ViacomCBS Excess 401(k) Plan. In no event shall any Reporting Employee be considered an Eligible Employee under the Plan.

2.15 The term "Employer" means the Company and any affiliate or subsidiary that adopts the Plan on behalf of its Eligible Employees, except as provided in Section 2.26.

2.16 The term "Investments Committee" means (i) for the periods prior to June 1, 2020, the Investments Committee for CBS Defined Contribution Plans, and (ii) for the periods on and after June 1, 2020, the ViacomCBS Investments Committee.

2.17 Prior to January 1, 2014 through December 31, 2014, the term "Investment Options" means the investment funds available to participants in the CBS 401(k) Plan, excluding the Self-Directed Brokerage Account and effective as of January 1, 2014, means the notional investment options selected by the CBS Retirement Committee in its sole discretion, and effective as of January 1, 2015, means the notional investment options elected by the Investments Committee, its sole discretion.

2.18 The term "Joint Payment Option" means the time and form of payment options available for the payment of an Account as described in Section 5.1.

2.19 The term "Joint Payment Option Election" means an election of a Joint Payment Option by a Participant as described in Section 5.1.

2.20 The term "Participant" means an Eligible Employee who elects to have Bonus Deferral Contributions made to the Plan.

2.21 The term "Payment Election" is defined in Section 5.1(a).

2.22 The term "Plan" means (i) for the periods prior to October 1, 2021, the CBS Bonus Deferral Plan, and (ii) for the periods on or after October 1, 2021, the ViacomCBS Bonus Deferral Plan. Part A of the Plan is attached hereto and shall apply to compensation which was Deferred prior to January 1, 2005. Part B of the Plan is set forth herein and shall apply to compensation which is Deferred on or after January 1, 2005. Certain provisions of this Part B apply as of certain earlier effective dates as specified herein. References to "the Plan" shall be considered references to Part A and/or Part B of the Plan as context requires.

2.23 The term "Post-2004 Subaccount" is defined in Section 4.1.

2.24 The term "Pre-2005 Subaccount" is defined in Section 4.1.

2.25 The term "Reporting Employee" is defined in Section 1.4.

2.26 The term "Separation from Service" means the condition that exists when a Participant and the Employer reasonably anticipate that no further services will be performed after a certain date or that the level of bona fide services that the Participant will perform after such date (whether as an employee or an independent contractor) would permanently decrease to no more than 20% of the average level of bona fide services performed (whether as an employee or an independent contractor) over the immediately preceding 36-month period (or the full period of services to the Employer if the Participant has been providing services to the Employer for less than 36 months). For purposes of this Section 2.26, for periods during which a Participant is on a paid bona fide leave of absence and has not otherwise experienced a Separation from Service, the Participant is treated as providing bona fide services at the level equal to the level of services that the Participant would have been required to perform to receive the compensation paid with respect to such leave of absence. Periods during which a Participant is on an unpaid bona fide leave of absence and has not otherwise experienced a Separation from Service are disregarded for purposes of this Section 2.26 (including for purposes of determining the applicable 36-month (or shorter) period). For purposes of this Section 2.26 and notwithstanding Section 2.15, the "Employer" shall be considered to include all members of the controlled group of corporations, trades or businesses which includes the Company; provided, however, that in applying Code Section 414(b), the phrase "at least 50 percent" shall be substituted for "at least 80 percent"; and in applying Code Section 414(c), the phrase "at least 50 percent" shall be used instead of the phrase "at least 80 percent." Separation from Service shall be determined on the basis of the modifications described in Treasury Regulation Section 1.409A-1 (or any successor regulation)) as defined in Code Section 409A and the regulations or other guidance issued thereunder.

2.27 The term "STIP" means (i) for the periods prior to October 1, 2021, the CBS Corporation Short-Term Incentive Plan, and (ii) for the periods on and after October 1, 2021, the ViacomCBS Senior Executive Short-Term Incentive Plan, as amended from time to time.

2.28 The term "Unforeseeable Emergency" means an event that results in severe financial hardship to a Participant resulting from (a) an illness or accident of the Participant or his or her spouse, dependent (as defined in Code Section 152, without regard to Code Sections 152(b)(1), (b)(2), and (d)(1)(B)), or beneficiary, (b) loss of the Participant's property due to casualty, or (c) other similar extraordinary circumstances arising due to results beyond the control of the Participant. This Section 2.28 shall be interpreted in a manner consistent with Code Section 409A and applicable provisions of the Treasury Regulations.

2.29 The term "ViacomCBS 401(k) Plan" means the ViacomCBS 401(k) Plan (formerly known as the CBS 401(k) Plan), originally effective as of September 1, 2001, and as amended from time to time thereafter (or any successor plan).

### Section 3. Participation.

- 3.1 *Election to Participate.* 
  - (a) An Eligible Employee must elect to participate in the Plan as provided below.

(b) To participate in the Plan for a calendar year, an Eligible Employee must make an annual election (a "Deferral Election") to defer receipt of a specified portion of his or her Bonus earned during such calendar year and scheduled to be paid in the succeeding calendar year in accordance with this Section 3. Subject to Section 3.1 such Deferral Election must be made not later than the Election Filing Date and shall be effective as of the Election Filing Date. For example, prior to December 31, 2009, an Eligible Employee may make a Bonus Deferral Contribution election with respect to any Bonus to be earned in 2010 that is scheduled to be paid in 2011. An Eligible Employee may make a Deferral Election whether or not such employee previously has made, or currently has in effect, any election to make Excess Salary Reduction Contributions under the CBS Excess 401(k) Plan, or its successor the ViacomCBS Excess 401(k) Plan. An Eligible Employee's entitlement to make Bonus Deferral Contributions shall cease with respect to any Bonus payable with respect to the calendar year following the calendar year in which he or she ceases to be an Eligible Employee.

(c) Notwithstanding the foregoing, an employee who first becomes an Eligible Employee during the course of a calendar year beginning on or after January 1, 2005 must make a Deferral Election with respect to any Bonus scheduled to be paid in the next succeeding calendar year within 30 days of the date he first becomes an Eligible Employee, provided that such employee has not already become eligible to participate in any other account balance plan of the Employer (as modified by Section 2.26). Such Deferral Election shall be effective on the date made and shall be effective with regard to the Bonus scheduled to be paid during the calendar year following the filing of the Deferral Election with the Administrative Committee, as determined pursuant to the pro-ration method permitted under Code Section 409A. If an Eligible Employee is a participant in another account balance plan that is required to be aggregated with the Plan under Code Section 409A when he first becomes eligible to participate in the Plan, such Eligible Employee shall be eligible to make a Deferral Election for the calendar year immediately following the calendar year of his initial eligibility by making an election in accordance with Section 3.1 (b) above.

(d) All Deferral Elections shall be made on a written or electronic form acceptable to the Administrative Committee (an "Election Agreement") filed with the Administrative Committee and shall specify the percentage of a Participant's Bonus that is to be deferred under the Plan during the applicable calendar year.

(e) All Deferral Elections relating to calendar years beginning on or after January 1, 2005, once effective, shall be irrevocable for that calendar year. All Participants are required to make a Deferral Election for each calendar year. If an Eligible Employee fails to make a Deferral Election for a given calendar year, the Eligible Employee shall not be entitled to participate in the Plan during that calendar year. Such Eligible Employee may resume participation in the Plan by completing and filing with the Administrative Committee a new Deferral Election by the Election Filing Date for the succeeding calendar year(s).

3.2 *Amount of Elections*. Each Deferral Election filed by an Eligible Employee must specify the amount of Bonus Deferral Contributions in a whole percentage between 1% and 15% of the Eligible Employee's applicable Bonus.

#### Section 4. Individual Account.

4.1 *Creation of Accounts.* The Company will establish and maintain on its books a reserve Account in the name of each Participant. Each Participant's Account will be credited with the amount of the Participant's Bonus Deferral Contributions (and earnings and losses thereon) made in all calendar years, including any Bonus Deferral Contributions for the Bonus earned in 2002 that are attributable to Deferral Elections originally made under the CBS Excess 401(k) Plan. A Participant's Account will be divided into the following subaccounts: (i) a "Pre-2005 Subaccount" for amounts Deferred as of December 31, 2004 (and earnings and losses thereon), and (ii) a "Post-2004 Subaccount" for amounts Deferred after December 31, 2004 (and earnings and losses thereon). Amounts in the Pre-2005 Subaccounts, which are intended to qualify for "grandfathered" status, shall be subject to the terms and conditions specified in Part A of the Plan as in effect as of October 3, 2004. A Participant will always be 100% vested in amounts credited to his or her Account hereunder.

#### 4.2 Investments.

(a) Prior to January 1, 2014, amounts, if any, in a Participant's Post-2004 Subaccount will be credited through December 31<sup>st</sup> of the calendar year in which the Participant experiences a Separation from Service with an amount equal to the amount which would have been earned had such amounts been invested in the same Investment Options and in the same proportion as the Participant may elect, from time to time, to have his contributions invested under (i) for the periods prior to October 1, 2021, the CBS 401(k) Plan, and (ii) for the periods on or after October 1, 2021, the ViacomCBS 401(k) Plan (other than the Self-Directed Account).

Effective as of January 1, 2014, a Participant may select from a list of notional Investment Options how the balance of his or her Account will be invested. If no selection is made, the Participant's Account will be notionally invested in the "qualified default investment alternative" within the meaning of (i) for the periods prior to October 1, 2021, the CBS 401(k) Plan, and (ii) for the periods on or after October 1, 2021, the ViacomCBS 401(k) Plan. Earnings and losses received on the Participant's notional investments will be credited to the Participant's Account in the manner designated by the Investments Committee. The Investments Descent and the reflection of value attributable to such notional investments in their Accounts, including, without limitation, procedures which restrict a Participant's ability to notionally invest in certain Investment Options.

(b) (i) Prior to October 2, 2017, when a Participant experienced a Separation from Service and elected (or was deemed to elect) to have his Post-2004 Subaccount distributed in a single lump sum, the Participant's Post 2004 Subaccount was credited with earnings based on the rate of return in the Fixed Income Fund (or any successor fund) beginning January 1<sup>st</sup> of the calendar year following the calendar year in which the Participant experienced a Separation from Service that resulted in the Participant's Post-2004 Subaccount becoming payable, and continuing through the date upon which such single lump sum payment was determined if such determination date was after December 31<sup>st</sup> of the calendar year in which the Participant experienced a Separation from Service. Payments due on January 31<sup>st</sup> of a calendar year were determined on the previous December 31<sup>st</sup>, while payments due on the first business day of a calendar month were determined on the last day of the second preceding calendar month (e.g., a payment scheduled for the first business day of March will be determined on the preceding January 31<sup>st</sup>).

(ii) Effective on and after October 2, 2017, a Participant who experienced or experiences a Separation from Service and elected (or is deemed to elect) to have his Post-2004 Subaccount distributed in a single lump shall have his Post-2004 Subaccount credited with earnings based on the rate of return in his selected Investment Options (or such Investment Options selected by the Investments Committee) until such time as the date upon which the single lump sum payment is determined and paid.

(c) (i) Prior to October 2, 2017, if a Participant experienced a Separation from Service and elected to have his Post-2004 Subaccount distributed in Annual Payments, the Participant's Post-2004 Subaccount were credited with earnings based on the rate of return in the Fixed Income Fund (or any successor fund) beginning January 1st of the calendar year following the calendar year in which the Participant experienced a Separation from Service that resulted in the Participant's Post-2004 Subaccount becoming payable, and continuing through the date upon which such Annual Payment was determined, if such determination date was after December 31st of the calendar year in which the Participant experienced a Separation from Service. Payments due on January 31st of a calendar year were determined on the previous December 31st, while payments due on the first business day of a calendar month were determined on the last day of the second preceding calendar month (e.g., a payment scheduled for the first business day of March was determined on the preceding January 31st).

(ii) Effective on and after October 2, 2017, a Participant who experienced or experiences a Separation from Service and elects to have his Post-2004 Subaccount distributed in Annual Payments shall have his Post-2004 Subaccount credited with earnings based on the rate of return in his selected Investment Options (or such Investment Options selected by the Investments Committee) until such time as the date upon which each Annual Payment is determined and paid.

(d) No provision of the Plan shall require the Company or the Employer to actually invest any amounts in any fund or in any other investment vehicle.

4.3 *Account Statements*. Each Participant will be given, at least annually, a statement showing (i) Bonus Deferral Contributions, and (ii) the balance of the Participant's Account after crediting Investments.

### Section 5. Payment.

5.1 Joint Payment Option Election.

(a) An Eligible Employee who has not elected or been deemed to have elected a Joint Payment Option under any other account balance plan that is required to be aggregated with the Plan under Code Section 409A shall, when he first becomes eligible to participate in the Plan, elect a Joint Payment Option on a written or electronic form acceptable to the Administrative Committee (a "Payment Election") at the same time that the Eligible Employee files his initial Deferral Election to commence participation in the Plan pursuant to Section 3.1 and in any event not later than his initial Election Filing Date. Such Payment Election shall be effective as of such initial Election Filing Date and shall be irrevocable. A Joint Payment Option elected pursuant to a Payment Election shall apply to all amounts credited to the Participant's Post-2004 Subaccount in the Plan and his Post-2004 Subaccount under any other account balance plan that is required to be aggregated with the Plan under Code Section 409A.

(b) (i) A Participant may elect to receive his entire Post-2004 Subaccount under either of the following Joint Payment Options: (A) a single lump sum; or, (B) annual payments over a period of two, three, four or five years ("Annual Payments"). The Annual Payments shall be treated as a single payment for purposes of this Section 5. If a Participant elects to receive Annual Payments over a period of two or more years, such Annual Payments shall be made in substantially equal annual payments, unless the Participant designates, at the time of making his Joint Payment Option Election, a specific percentage of his Post-2004 Subaccount to be distributed in each year. All specified percentages must be a whole multiple of and the total of all designated percentages must be equal to 100%. Effective as of October 2, 2017, if a Participant elects to receive Annual Payments, and the Participant shall not be able to designate a specific percentage of his Post-2004 Subaccount to be distributed in each years.

(ii) If a Participant makes a Joint Payment Option Election to receive Annual Payments, the first payment shall be made on the later of (A) January 31<sup>st</sup> of the calendar year immediately following the calendar year in which the Participant experiences a Separation from Service or (B) the first business day of the seventh calendar month following the calendar month in which the Participant experiences a Separation from Service, and any subsequent Annual Payments shall be made on each applicable January 31<sup>st</sup> thereafter.

(iii) If a Participant makes a Joint Payment Option Election to receive payments in a single lump sum, such lump sum payment shall be made on the later of (A) January  $31^{st}$  of the calendar year immediately following the calendar year in which the Participant experiences a Separation from Service or (B) the first business day of the seventh calendar month following the calendar month in which the Participant experiences a Separation from Service. Alternatively, a Participant may elect for the single lump sum to be paid on January  $31^{st}$  of the second, third, fourth, or fifth calendar year following the end of the calendar year in which the Participant experiences a Separation from Service.

(iv) If a Participant does not make a Joint Payment Option Election in accordance with the terms of the Plan or under any other account balance plan that is required to be aggregated with the Plan under Code Section 409A, such Participant shall be deemed to have made a Joint Payment Option Election to receive his Post-2004 Subaccount in a single lump sum payable in accordance with the first sentence of Section 5.1(b)(iii).

(v) The following examples illustrate the provisions of this Section:

<u>Example 1</u>: Assume that a Participant elects (or is deemed to elect) a Joint Payment Option that provides for a single lump sum payment on the later of (A) January 31<sup>st</sup> of the calendar year following the calendar year in which he incurs a Separation from Service or (B) the first business day of the seventh calendar month following the calendar month in which the Participant experiences a Separation from Service, and the Participant experiences a Separation from Service on March 15, 2021. The lump sum shall be paid on January 31, 2022. The Participant alternatively could have elected to receive his lump sum payment on January 31, 2023, 2024, 2025 or 2026.

Example 2: Same facts as Example 1, except the Participant experiences a Separation from Service on September 15, 2021. In this example, the lump sum will be paid on the first business day in April 2022.

Example 3: If a Participant elects a Joint Payment Option that provides for Annual Payments over a period of four years in the event of a Separation from Service and experiences a Separation from Service on March 15, 2021, each payment on January 31, 2022 through 2025 will be comprised of approximately 25% of the Participant's Post-2004 Subaccount as of the Participant's date of Separation from Service, though the actual amount of each payment may not be the same due to crediting of investment gains and losses through December 31<sup>st</sup> of the calendar year prior to the calendar year of each such payment, or, effective after October 2, 2017, the actual amount of each payment may not be the same due to crediting of investment gains and losses through the payment date . Prior to October 2, 2017, a Participant alternatively could designate that 10% of his Post-2004 Subaccount be distributed on January 31, 2010, 20% on January 31, 2011, 30% on January 31, 2012 and 40% on January 31, 2013, or, any other combination of percentages that totals 100%.

Example 4: Same facts as Example 3, except the Participant experiences a Separation from Service on September 15, 2021. In this example, the first payment shall be made on the first business day in April 2022, and the remaining three payments will be made on January 31, 2023, 2024 and 2025. The alternative schedule described in Example 3 would result in payment of 10% of his Post-2004 Subaccount on the first business day in April 2022, 20% on January 31, 2023, 30% on January 31, 2024 and 40% on January 31, 2025.

5.2 *Payment on Account of Separation from Service.* If a Participant experiences a Separation from Service prior to his death, the Participant shall commence receiving payments from his Post-2004 Subaccount in accordance with the Joint Payment Option Election in effect with respect to the Participant.

5.3 *Payment on Account of Participant's Death.* If a Participant dies prior to his Separation from Service, or after his Separation from Service but prior to the distribution of his entire Post-2004 Subaccount, the Participant's entire Post-2004 Subaccount shall be paid to the Participant's beneficiary in a single lump sum payment within 90 days of the Participant's death. The Participant's Post-2004 Subaccount shall continue to be credited with earnings in accordance with Section 6.2 until his entire Post-2004 Subaccount is distributed.

### Section 6. Unforeseeable Emergency Distributions and Deferral Revocations.

A Participant may request the Administrative Committee to accelerate distribution of all or any part of the value of his Post-2004 Subaccount solely for the purpose of alleviating an Unforeseeable Emergency. Payments of amounts as a result of an Unforeseeable Emergency may not exceed the amount necessary to satisfy such Unforeseeable Emergency, plus amounts necessary to pay taxes reasonably anticipated as a result of the distribution, and after taking into account any additional compensation that is available to the Participant upon cancellation of the Participant's Bonus Contributions. The Administrative Committee may request that the Participant provide certifications and other evidence of qualification for such Unforeseeable Emergency distribution as it determines appropriate. The decision of the Administrative Committee with respect to the grant or denial of all or any part of such request shall be in the sole discretion of the Administrative Committee, even if the Participant demonstrates that an Unforeseeable Emergency exists, and shall be final and binding and not subject to review. If a Participant receives a distribution upon an Unforeseeable Emergency pursuant to this Section 6, or a hardship withdrawal under the ViacomCBS 401(k) Plan (or its predecessor, the CBS 401(k) Plan, as applicable), the Participant's Deferral Election will be canceled in its entirety for the remainder of the calendar year in which such Unforeseeable Emergency distribution is made under the Plan and under any other account balance plan that is required to be aggregated with the Plan under Code Section 409A.

### Section 7. Beneficiary Designation.

A Participant's beneficiary designation for the Plan will automatically be the same as the Participant's beneficiary designation recognized under the ViacomCBS 401(k) Plan (or its predecessor, the CBS 401(k) Plan, as applicable) unless a separate written designation of beneficiary form for the Plan has been properly filed with the Administrative Committee in a form acceptable to the Administrative Committee. In the absence of such a designation and at any other time when there is no existing beneficiary designated hereunder, the beneficiary of the Participant for payment of his Post-2004 Subaccount hereunder shall be the estate of the Participant. If two or more persons designated as a Participant's beneficiary are in existence with respect to his Post 2004 Subaccount, the amount of any lump sum payment payable hereunder shall be divided equally among such persons unless the Participant's beneficiary designation specifically provides for a different allocation.

### Section 8. Nature of Interest of Participant.

Participation in the Plan will not create, in favor of any Participant, any right or lien in or against any of the assets of the Company or any Employer, and all amounts of compensation deferred hereunder shall at all times remain an unrestricted asset of the Company or the Employer. A Participant's rights to benefits payable under the Plan are not subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, or encumbrance. All payments hereunder shall be paid in cash from the general funds of the Company or applicable Employer and no special or separate fund shall be established and no other segregation of assets shall be made to assure the payment of benefits hereunder. Nothing contained in the Plan, and no action taken pursuant to its provisions, shall create or be construed to create a trust of any kind, or a fiduciary relationship, between any Employer and a Participant or any other person, and the Company's and each Employer's promise to pay benefits hereunder shall at all times remain unfunded as to the Participant.

### Section 9. Administration.

9.1. Administrative Committee. The Plan shall be administered by the Administrative Committee. The Administrative Committee shall have sole and absolute discretion to interpret, where necessary, the provisions of the Plan (including, without limitation, by supplying omissions from, correcting deficiencies in, or resolving inconsistencies or ambiguities in, the language of the Plan), to determine the rights and status under the Plan of any Participant and any other persons, to resolve questions or disputes arising under the Plan and to make any determinations with respect to the benefits hereunder and the persons entitled thereto as may be necessary for the purposes of the Plan.

9.2 *Powers of the Administrative Committee.* In furtherance of, but without limiting, Section 9.1, the Administrative Committee shall have the following specific authorities, which it shall discharge in its sole and absolute discretion in accordance with the terms of the Plan (as interpreted, to the extent necessary, by the Administrative Committee):

(i) to determine who are Eligible Employees for purposes of participation in the Plan;

(ii) to interpret the terms and provisions of the Plan and to determine any and all questions arising under the Plan, including without limitation, the right to remedy possible ambiguities, inconsistencies, or omissions by a general rule or particular decision;

- (iii) to adopt rules consistent with the Plan;
- (iv) to approve certain amendments to the Plan;
- (v) to determine the amounts payable to any person under the Plan; and
- (vi) to conduct the claims procedure specified in Section 9.3.

9.3 *Claims Procedure.* The Administrative Committee shall have the exclusive right to interpret the Plan and to decide any and all matters arising thereunder. In the event of a claim by a Participant as to the amount of any distribution or method of payment under the Plan, within 90 days of the filing of such claim, unless special circumstances require an extension of such period, such person will be given notice in writing of any denial, which notice will set forth the reason for the denial, the Plan provisions on which the denial is based, an explanation of what other material or information, if any, is needed to perfect the claim, and an explanation of the claims review procedure. The Participant may request a review of such denial within 60 days of the date of receipt of such denial by filing notice in writing with the Administrative Committee. The Participant will have the right to review pertinent Plan documents and to submit issues and comments in writing. The Administrative Committee will respond in writing to a request for review within 60 days of receiving it, unless special circumstances require an extension of such period. The Administrative Committee, at its discretion, may request a meeting to clarify any matters deemed appropriate.

9.4 *Finality of Administrative Committee Determinations*. Determinations by the Administrative Committee and any interpretation, rule, or decision adopted by the Administrative Committee under the Plan or in carrying out or administering the Plan shall be final and binding for all purposes and upon all interested persons, their heirs, and personal representatives.

### Section 10. No Employment Rights.

No provisions of the Plan or any action taken by the Company, any Employer, the Board of Directors, or the Administrative Committee shall give any person any right to be retained in the employ of the Company or any Employer, and the right and power of the Company or any Employer to dismiss or discharge any Participant is specifically reserved.

### Section 11. Amendment, Suspension, and Termination.

The Administrative Committee shall have the right to amend the Plan at any time, unless provided otherwise in the Company's governing documents. The Board of Directors shall have the right to suspend or terminate the Plan at any time. No amendment, suspension or termination shall, without the consent of a Participant, adversely affect such Participant's rights in his Account; provided, however, that the consent requirement of Participants to certain actions shall not apply to any amendment or termination that is deemed necessary by the Company to avoid the imposition on any person of additional taxes, penalties or interest under Code Section 409A. In the event the Plan is terminated, the Administrative Committee may continue to administer the Plan in accordance with the relevant provisions thereof or shall have the right to change the time and form of distribution of Participants' Accounts, including requiring that the Accounts be immediately distributed in the form of a lump sum payment; provided, however, that no such change in the time or form of payment shall cause the Plan to fail to comply with the requirements of Code Section 409A.

#### Section 12. Miscellaneous.

12.1 *Severability*. If a provision of the Plan shall be held invalid, the invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the invalid provision had not been included in the Plan.

12.2 *Governing Law.* The provisions of the Plan shall be governed by and construed in accordance with the laws of the State of New York, to the extent not preempted by the laws of the United States.

12.3 *Gender*. Wherein used herein, words in the masculine form shall be deemed to refer to females as well as males.

12.4 *Code Section 409A.* To the extent applicable, it is intended that the Plan comply with the provisions of Code Section 409A. References to Code Section 409A shall include any proposed, temporary or final regulation, or any other guidance, promulgated with respect to such section by the U.S. Department of the Treasury or the Internal Revenue Service. The Plan shall be administered and interpreted in a manner consistent with this intent. If any provision of the Plan is susceptible of two interpretations, one of which results in the compliance of the Plan with Code Section 409A and the applicable Treasury Regulations, and one of which does not, then the provision shall be given the interpretation that results in compliance with Code Section 409A and the applicable Treasury Regulations. Notwithstanding the foregoing or any other provision of the Plan to the contrary, neither the Company nor any of its subsidiaries or affiliates shall be deemed to guarantee any particular tax result for any Participant, spouse, or beneficiary with respect to any payments provided hereunder.

IN WITNESS WHEREOF, in accordance with the Administrative Committee's August 20, 2021 Unanimous Written Consent, the Company has caused this Plan to be executed by its duly authorized representative this 20th day of September, 2021.

ViacomCBS Inc.

By:	/s/ Mark Beatty
	Mark Beatty

Its: <u>Member, ViacomCBS Administrative Committee</u>

#### AMENDMENT NO. 1 TO THE AMENDED AND RESTATED VIACOMCBS BONUS DEFERRAL PLAN-PART A (2021 RESTATEMENT) & PART B (2021 RESTATEMENT)

Effective as of February 16, 2022, the ViacomCBS Bonus Deferral Plan (each of Part A and Part B) is amended such that, for all periods on or after February 16, 2022, all references to "ViacomCBS" or "ViacomCBS Inc." shall be deemed to refer to "Paramount Global", including such that: all references to the "ViacomCBS Administrative Committee" or "ViacomCBS Investments Committee" shall be deemed to refer to "Paramount Global Administrative Committee" and "Paramount Global Investments Committee", respectively; all references to "ViacomCBS 401(k) Plan" shall be deemed to refer to "Paramount Global 401(k) Plan"; and reference to "ViacomCBS Inc." in the definition of "Company" shall be deemed to refer to "Paramount Global".

IN WITNESS WHEREOF, pursuant to a resolution of the Paramount Global Administrative Committee, the undersigned hereby executes this amendment this 7<sup>th</sup> day of October, 2022.

By: /s/ Mark Beatty

Name: Mark Beatty

Its: Member, Paramount Global Administrative Committee

#### AMENDMENT NO. 1 TO THE AMENDED AND RESTATED VIACOMCBS BONUS DEFERRAL PLAN FOR DESIGNATED SENIOR EXECUTIVES-PART A (2021 RESTATEMENT) & PART B (2021 RESTATEMENT)

Effective as of February 16, 2022, the ViacomCBS Bonus Deferral Plan for Designated Senior Executives (each of Part A and Part B) is amended such that, for all periods on or after February 16, 2022, all references to "ViacomCBS" or "ViacomCBS Inc." shall be deemed to refer to "Paramount Global", including such that: all references to the "ViacomCBS Administrative Committee" or "ViacomCBS Investments Committee" shall be deemed to refer to "Paramount Global Investments Committee", respectively; all references to "ViacomCBS 401(k) Plan" shall be deemed to refer to "Paramount Global 401(k) Plan"; and reference to "ViacomCBS Inc." in the definition of "Company" shall be deemed to refer to "Paramount Global".

IN WITNESS WHEREOF, pursuant to a resolution of the Paramount Global Administrative Committee, the undersigned hereby executes this amendment this 7<sup>th</sup> day of October, 2022.

By: /s/ Mark Beatty

Name: Mark Beatty

Its: Member, Paramount Global Administrative Committee

#### SHEARMAN & STERLING LLP

#### 599 LEXINGTON AVENUE NEW YORK, NY 10022-6069 +1.212.848.4000

October 7, 2022

Paramount Global 1515 Broadway New York, NY 10036 (212) 258-6000

Ladies and Gentlemen:

We are acting as counsel for Paramount Global, a Delaware corporation (the "<u>Company</u>"), in connection with the preparation and filing by the Company of a registration statement on Form S-8 (the "<u>Registration Statement</u>") with the Securities and Exchange Commission (the "<u>Commission</u>") under the Securities Act of 1933, as amended (the "<u>Securities Act</u>"), with respect to registration of additional \$150,000,000 of unsecured obligations of the Company to pay deferred compensation in the future (the "<u>Deferred Compensation Obligations</u>") in accordance with the terms of the Paramount Global Excess 401(k) Plan (f/k/a the ViacomCBS Excess 401(k) Plan, Paramount Global Excess, Paramount Global Bonus Deferral Plan (f/k/a the ViacomCBS Bonus Deferral Plan) and the Paramount Global Bonus Deferral Plan for Designated Senior Executives (f/k/a the ViacomCBS Bonus Deferral Plan for Designated Senior Executives), Collectively, the "<u>Plans</u>").

In connection with the foregoing, we have reviewed originals or copies identified to our satisfaction of the following documents:

- (a) The Registration Statement;
- (b) The certificate of incorporation and by-laws of the Company, in each case as amended; and
- (c) Originals or copies of such other corporate records of the Company, certificates of public officials and of officers of the Company, and agreements and other documents as we have deemed necessary as a basis for the opinions expressed below.

In our examination, we have assumed the genuineness of all signatures, the authenticity of all documents, certificates and instruments submitted to us as originals and the conformity with originals of all documents submitted to us as copies.

Our opinion set forth below is based on the text of the Plans as referenced in the Exhibit Index to the Registration Statement.

Subject to the foregoing and having regard for such legal considerations as we have deemed relevant, it is our opinion that the Deferred Compensation Obligations, when incurred in accordance with the terms of the Plan, will constitute legal, valid and binding obligations of the Company, enforceable against the Company in accordance with their terms and terms of the of the Plan, except as enforceability (i) may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting creditor's rights generally, and (ii) is subject to general principals of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

#### SHEARMAN.COM

Shearman & Sterling LLP is a limited liability partnership organized in the United States under the laws of the state of Delaware, which laws limit the personal liability of partners.

We are opining herein as to the General Corporation Law of the State of Delaware, and we express no opinion with respect to any other laws.

This opinion letter speaks only as of the date hereof. We expressly disclaim any responsibility to advise you of any development or circumstance of any kind, including any change of law or fact that may occur after the date of this opinion letter that might affect the opinions expressed herein.

We hereby consent to the filing of this opinion as an exhibit to the Registration Statement. In giving this consent, we do not thereby concede that we come within the category of persons whose consent is required by the Securities Act or the General Rules and Regulations of the Commission promulgated thereunder.

This opinion is rendered solely to you in connection with the above matter. This opinion may not be relied upon by you for any other purpose without our prior written consent.

Very truly yours,

/s/ Shearman & Sterling LLP

Shearman & Sterling LLP

#### SHEARMAN.COM

Shearman & Sterling LLP is a limited liability partnership organized in the United States under the laws of the state of Delaware, which laws limit the personal liability of partners.

### CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We hereby consent to the incorporation by reference in this Registration Statement on Form S-8 of Paramount Global of our report dated February 15, 2022 relating to the financial statements, financial statement schedule and the effectiveness of internal control over financial reporting, which appears in Paramount Global's Annual Report on Form 10-K for the year ended December 31, 2021.

/s/ PricewaterCoopers LLP

New York, New York October 7, 2022

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 3<sup>rd</sup> day of October, 2022.

<u>/s/ Barbara M. Byrne</u> Name: Barbara M. Byrne

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 28<sup>th</sup> day of September, 2022.

<u>/s/ Candace K. Beinecke</u> Name: Candace K. Beinecke

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 27<sup>th</sup> day of September, 2022.

<u>/s/ Charles E. Phillips, Jr.</u> Name: Charles E. Phillips, Jr.

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 27<sup>th</sup> day of September, 2022.

<u>/s/ Frederick O. Terrell</u> Name: Frederick O. Terrell

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 27<sup>th</sup> day of September, 2022.

<u>/s/ Judith A. McHale</u> Name: Judith A. McHale

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 21<sup>st</sup> day of September, 2022.

<u>/s/ Linda Griego</u> Name: Linda M. Griego

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 21<sup>st</sup> day of September, 2022.

<u>/s/ Nicole Seligman</u> Name: Nicole Seligman

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 27<sup>th</sup> day of September, 2022.

<u>/s/ Robert N. Klieger</u> Name: Robert N. Klieger

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 27<sup>th</sup> day of September, 2022.

<u>/s/Ronald L. Nelson</u> Name: Ronald L. Nelson

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 27<sup>th</sup> day of September, 2022.

<u>/s/ Shari Redstone</u> Name: Shari E. Redstone

KNOW ALL PERSONS BY THESE PRESENTS that the undersigned hereby constitutes and appoints Christa A. D'Alimonte as attorney-in-fact, with the power of substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign the Registration Statement on Form S-8 (the "Registration Statement") currently contemplated to be filed on October 7, 2022, and any and all amendments (including post-effective amendments) to the Registration Statement and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes she might or could do in person, hereby ratifying and confirming all that such attorney-in-fact and agent or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has caused this Power of Attorney to be executed as of this 27<sup>th</sup> day of September, 2022.

<u>/s/ Susan Schuman</u> Name: Susan Schuman

## CALCULATION OF FILING FEE TABLE

**S-8** 

# (Form Type)

# PARAMOUNT GLOBAL

(Exact Name of Registrant as Specified in its Charter)

## Table 1: Newly Registered Securities

Security Type	Security Class Title	Fee Calculation Rule	Amount Registered	Proposed Maximum Offering Price Per Unit	Maximum Aggregate Offering Price	Fee Rate	Amount of Registration Fee
Debt	Deferred Compensation Obligations (1)	Other	\$110,000,000	100%	\$110,000,000 (2)	0.00011020	\$12,122
Debt	Deferred Compensation Obligations (3)	Other	\$40,000,000	100%	\$40,000,000 (2)	0.00011020	\$4,408
Total Offering Amounts				\$150,000,000		\$16,530	
Total Fee Offsets							\$0
Net Fee Due							\$16,530

(1) Amount of Deferred Compensation Obligations registered with respect to the Paramount Global Excess 401(k) Plan and the Paramount Global Excess 401(k) Plan for Designated Senior Executives.

(2) Solely for the purposes of computing the registration fee pursuant to Rule 457(h) under the Securities Act of 1933, as amended (the "Securities Act"), the amount of deferred compensation obligations registered is based on an estimate of the amount of compensation participants may defer under the Plans.

(3) Amount of Deferred Compensation Obligations registered with respect to the Paramount Global Bonus Deferral Plan and the Paramount Global Bonus Deferral Plan for Designated Senior Executives.